

Denoising and segmentation in medical image analysis: A comprehensive review on machine learning and deep learning approaches

Ravi Ranjan Kumar¹ · Rahul Priyadarshi²

Received: 29 January 2024 / Revised: 19 April 2024 / Accepted: 29 April 2024 © The Author(s), under exclusive licence to Springer Science+Business Media, LLC, part of Springer Nature 2024

Abstract

Medical imaging plays an essential role in modern healthcare, helping accurate diagnoses and effective treatment strategies. Still, the quality and interpretability of medical images are regularly hindered by various sources of noise. This paper presents a comprehensive exploration of traditional noise reduction techniques in medical imaging, addressing challenges posed by quantum noise, electronic noise, radiation interference, and other factors. The study delves into spatial filtering, frequency domain filtering, statistical methods, probability-based noise reduction, and adaptive filtering techniques. Each method is analyzed for its applicability and effectiveness in mitigating noise while preserving diagnostically relevant information. The comparative analysis provides insights into the strengths and limitations of these techniques, guiding practitioners in selecting appropriate methods based on imaging modalities and noise characteristics. Also, the paper highlights future research directions, emphasizing the potential of advanced Machine Learning (ML) models and the integration of multimodal data for enhanced noise removal.

Keywords Medical imaging \cdot Noise \cdot Reduction \cdot Machine learning \cdot Modalities \cdot Image quality \cdot Diagnostic

1 Introduction

Medical imaging has emerged as an essential aspect of modern healthcare, influencing the detection, treatment, management, and monitoring of a variety of diseases. The field combines modern equipment to develop comprehensive visual models of the human body's internal structure, helping healthcare professionals to investigate and assess both normal

Ravi Ranjan Kumar kravirrk@gmail.com

Rahul Priyadarshi rahul.glorious91@gmail.com

¹ National Institute of Technology, Patna, Bihar 800005, India

² Faculty of Engineering and Technology, ITER, Siksha 'O' Anusandhan (Deemed to Be University), Bhubaneswar 751030, India

physiological functioning and abnormal anomalies [1, 2]. The importance of medical imaging in the healthcare system is enormous, with wide-ranging consequences that extend beyond conventional medicine. The primary benefit of this technology is its ability to visualize what was previously undetected, allowing us to peer through the skin and observe the intricate structures of the human body. Medical imaging provides an innovative insight into the intricate structure of an infant's developing heart and the effects of cancer in a patient's lung [3, 4]. Medical imaging is crucial for enabling the early identification and diagnosis of illnesses. Medical experts may diagnose illnesses in their early stages using imaging modalities such as X-rays, Computed Tomography (CT) scans, Magnetic Resonance Imaging (MRI), ultrasounds, and Positron Emission Tomography (PET) scans, commonly before symptoms occur. Mammography plays an important part in detecting breast cancer at an early stage, increasing the chances of successful treatment, and saving lives. Lung cancer screening with CT scans may detect lung cancers in their early stages, providing an opportunity for intervention [5, 6]. Medical imaging supervises the whole healthcare process, from treatment planning to exact implementation. Surgeons may meticulously organize intricate surgeries to minimize damage to healthy tissue. Real-time imaging, including fluoroscopy, assists in directing surgical tools and evaluating progress throughout surgery. Radiation therapy and chemotherapy may be precisely targeted at tumor sites, minimizing side effects, and improving therapeutic effectiveness. Medical imaging is essential for monitoring disease development and analyzing the effectiveness of treatments, as well as playing a significant role in diagnosing, treating, and detecting illnesses [7, 8]. Regular scans are crucial for overseeing chronic conditions like cancer or cardiovascular disease since they enable healthcare providers to track the progression of the illness and assess the efficacy of therapies. Early improvements to treatment protocols according to diagnostic information could improve outcomes for patients.

Also, there is constant research and development in the field of medical imaging. Advanced technologies, including 3D imaging, functional MRI, and molecular imaging, continue to expand the boundaries of medical knowledge. The advancements improve our knowledge of disease processes, help create new medicines, and boost diagnostic procedures. The importance of noise reduction and segmentation in medical image analysis is crucial [9, 10]. Noise from many sources during picture collection might distort crucial features and impede diagnostic accuracy. Reducing such noise is essential not just to achieve visual clarity but also to ensure the effectiveness of automated image analysis techniques. Segmentation, which involves separating an image into important components, is also essential [11, 12]. Segmentation is essential, not only for diagnosing but also for modifying therapies to fit individual patient characteristics. But challenges persist, which include the variation of noise across different imaging techniques, physical variations among people, and the limited availability of accurate data to use for model training [13].

Advanced technologies, such as Deep Learning (DL), provide interesting responses in this dynamic environment. Convolutional Neural Networks (CNNs) are very effective in noise reduction tasks due to their capacity to adapt to the several noise characteristics present across various modalities. DL models like U-Net structures work well for segmentation because they find a good balance between accurate localization and low computational cost [14, 15]. Combining data from multiple domains enhances the overall analysis by incorporating information from various sources. Despite progress, the need for explainable AI is essential to maintaining openness in algorithmic decision-making, particularly in medical fields where trust and comprehension are essential. Real-time segmentation is an innovative factor that improves workflow efficiency and enables dynamic treatment techniques. Processing medical images in real-time speeds up diagnosis, particularly in emergency situations, and helps with adaptive therapies by providing immediate insight into evolving clinical conditions [16, 17].

Finally, the collaboration of medical imaging, noise reduction, and segmentation has advanced healthcare into an innovative era of accuracy and effectiveness. The exploration continues, including historical landmarks and the incorporation of advanced technology. Challenges remain, but each obstacle provides opportunities for innovation and collaboration [18, 19]. As we proceed ahead, it is important to not only emphasize technology developments but also integrate them into clinical processes to ensure that medical imaging can fully enhance patient outcomes. The commitment to creativity, collaboration, and an indepth knowledge of the intersection of medical imaging, noise reduction, and segmentation remains the main objective in the achievement of excellence in healthcare.

This work aims to improve the efficiency of medical image analysis by overcoming the issues of noise and segmentation. Various sources of noise, such as acquisition techniques and environmental conditions, may cause ambiguities that may prevent an accurate diagnosis [20, 21]. Segmentation, a method of splitting an image into significant parts, is essential for distinguishing specific characteristics or irregularities. Thus, it is crucial to understand and reduce noise while enhancing segmentation methods to progress in medical image analysis.

2 Research objectives

The objectives of the paper are as follows:

- Provide a comprehensive review of modern approaches for noise reduction and segmentation in medical imaging.
- Analyze traditional techniques utilized for noise reduction and image segmentation.
- Compare traditional techniques with modern machine learning (ML) algorithms for noise reduction and segmentation.
- Provide concepts on the suitability and effectiveness of these methods in various medical imaging applications.
- Investigate and evaluate traditional noise reduction methods used in medical imaging.
- Study spatial filtering, frequency domain filtering, statistical approaches, probabilitybased noise reduction, and adaptive filtering techniques.
- Provide guidance to professionals on choosing suitable methods according to imaging modalities and noise factors.
- Study how advanced ML models and the use of multimodal data may improve noise reduction in medical imaging.
- Identify potential research approaches to enhance noise reduction techniques in medical imaging applications.

2.1 Paper organization

In order to achieve these objectives, the paper is organized in a way that facilitates a comprehensive and systematic analysis of various issues related to the removal of noise and segmentation in medical imaging. The following sections provide an organized structure for this paper, working as an overview for readers to navigate the comprehensive review that is being presented.

- Section 3 provides a basic introduction to essential medical imaging modalities and emphasizes their important role in healthcare.
- Section 4 explores the complexities of noise reduction, including both conventional methods and the recent development of ML-based approaches.
- Section 5 discusses image segmentation, highlighting its importance and providing information on both conventional and advanced ML-based techniques.
- Section 6 discusses the challenges involved in noise reduction and segmentation, recognizing variations across imaging techniques, anatomical changes, and the obstacles of obtaining accurate data for comparison.
- The study in Section 7 discusses future research areas such as improved ML models, multimodal data fusion, explainable artificial intelligence, and real-time segmentation.
- The conclusion in Section 8 summarizes the main findings, emphasizing their impact on medical imaging and emphasizing the need for efficient noise reduction and segmentation in improving healthcare outcomes.

The article provides an overview for readers to help them navigate the intricate area and understand the constantly evolving terrain of medical imaging.

3 Medical imaging in healthcare: A visual revolution

Medical imaging plays an essential role in modern medical treatment by presenting the hidden complexities of the human body with excellent clarity. This overview presents a wide range of imaging techniques, from conventional X-rays to advanced techniques such as MRI and CT scans. We investigate how medical imaging improves diagnosis, optimizes medical treatment, and encourages us toward precision medicine [22, 23]. Simply put, it is a visual revolution that is transforming how medical professionals perceive and improve patient outcomes.

3.1 Types of medical imaging modalities

Medical imaging methods play an important role in providing specialists with novel insights into the human body within the dynamic healthcare sector. This section provides an analysis of several imaging modalities, each with unique features adapted to specific diagnostic requirements [24, 25]. From X-rays showing skeletal structures to CT scans providing cross-sectional imaging and MRIs identifying soft tissue, these modalities collectively give a complete toolset. Recognizing the principles and uses of X-ray, MRI, CT, and other modalities is essential for understanding their significance in diagnostic techniques and therapeutic treatments [26, 27].

The above Table 1 provides a quick overview of diagnostic imaging modalities, summarizing their principles, applications, significance in clinical practice, noise types, distribution characteristics, and radiation sources. This overview serves as a quick reference for medical professionals and scholars seeking a comprehensive understanding of diagnostic imaging techniques.

Table 1 Medical imaging n	nodalities overview				
Medical Imaging Modali- ties	Types of Noise Distribu- tion	Radiation Source	Principle	Applications	Significance
X-ray	Quantum Noise, Electronic Noise, Scintillation Noise	X-Ray Tube	Differential absorption of X-ray photons by tissues	Fracture detection, chest abnormalities, dental examinations	Quick and effective imaging, foundational for bones
MRI	Thermal Noise, System Noise, Motion Artifacts	Magnetic Fields	Interaction of hydrogen nuclei with a magnetic field and RF pulses	Brain, spinal cord, joints, soft tissues imaging	Exceptional soft tissue con- trast, crucial for neurology
ст	Electronic Noise, Beam Hardening, Scatter Radiation	X-Ray Tube, Detectors	Combination of X-ray technology with com- puter processing	Detailed examinations of internal structures, tumor detection	Cross-sectional, 3D images for precise diagnostics
Ultrasound	Speckle Noise, Electronic Noise, Acoustic Noise	Transducer	Utilization of high-fre- quency sound waves for real-time imaging	Obstetrics, cardiology, abdominal imaging	Non-invasive, real-time imaging, versatile
Nuclear Medicine	Photon Noise, Electronic Noise, Scatter Radiation	Radioactive Tracers	Administration of radioac- tive tracers emitting gamma rays	Myocardial perfusion imaging, PET for cancer staging, neurological disorders	Functional insights into organ function and metabolism
PET	Coincidence Timing Error, Scatter Radiation	Positron-Emitting Radio- nuclides	Injection of a radioac- tive substance emitting positrons	Oncology, cardiology, neurology	Functional images reflecting metabolic activity
Mammography	Quantum Noise, Electronic Noise, Motion Artifacts	X-Ray Tube	Use of X-rays to create detailed images of breast tissue	Breast cancer screening and diagnosis	Primary tool for early detec- tion of breast cancer
Fluoroscopy	Quantum Noise, Electronic Noise, Motion Artifacts	X-Ray Tube	Continuous X-ray imaging for real-time visualiza- tion of moving structures	Barium studies, catheter placements, joint injec- tions	Instrumental in interven- tional procedures, real- time guidance

3.1.1 X-ray imaging

X-ray imaging, a pioneering modality discovered by Wilhelm Roentgen in 1895, relies on the principle of differential absorption of X-ray photons by tissues. This modality excels at visualizing dense structures like bones due to their higher X-ray absorption [28, 29]. Widely employed in diagnostic radiology, X-ray imaging provides a rapid and effective means for detecting fractures, assessing orthopedic conditions, and uncovering abnormalities within the chest, including pneumonia and lung cancer. Its versatility extends to dental examinations, where it aids in the prompt diagnosis and planning of interventions for oral health issues. The enduring significance of X-ray imaging lies in its accessibility, speed, and critical role in providing essential diagnostic information across various medical disciplines.

3.1.2 Magnetic Resonance Imaging (MRI)

MRI, a transformative technology, harnesses the behavior of hydrogen nuclei in response to magnetic fields and radiofrequency pulses. This non-invasive technique captures detailed, high-contrast images of soft tissues by processing the signals generated during the realignment of hydrogen nuclei [30, 31]. Widely acclaimed for its exceptional soft tissue contrast, MRI plays an indispensable role in medical imaging. In neurology, it stands as a key diagnostic tool for conditions such as multiple sclerosis, offering unparalleled insights into the intricacies of the brain and spinal cord. Moreover, its applications extend to musculoskeletal imaging, facilitating precise assessments of ligaments, tendons, and cartilage, essential in orthopedic evaluations. Beyond its neuro and musculoskeletal provess, MRI's versatility spans various medical disciplines, establishing it as a cornerstone for comprehensive and detailed diagnostic imaging throughout the body [32].

3.1.3 Computed Tomography (CT) Imaging

CT stands at the intersection of X-ray technology and advanced computer processing, facilitating the creation of intricate cross-sectional images with three-dimensional precision. This modality employs a rotating X-ray source and detectors, capturing detailed snapshots of internal structures. CT imaging's versatility is represented in its widespread use for comprehensive examinations of the head, chest, abdomen, and pelvis [33, 34]. Its diagnostic prowess extends to the identification of conditions such as tumors, vascular abnormalities, and traumatic injuries, offering unparalleled clarity in delineating anatomical details. In the realm of vascular imaging, CT angiography plays a pivotal role, providing detailed assessments of blood vessels and enabling the detection of vascular diseases. CT imaging's ability to deliver high-resolution, multi-dimensional images positions it as an indispensable tool in the diagnostic arsenal, contributing significantly to accurate diagnosis and treatment planning across diverse medical specialties.

3.1.4 Ultrasound imaging

Ultrasound imaging, harnessing high-frequency sound waves, offers real-time visualizations of internal structures by processing echoes generated during their interaction. Its noninvasive nature and dynamic imaging capabilities render ultrasound a versatile modality across various medical domains. Widely embraced in obstetrics, it becomes a vital tool for monitoring fetal development, providing expectant parents and healthcare professionals with valuable insights. In cardiology, ultrasound serves as a cornerstone for assessing heart function, capturing detailed images that aid in the diagnosis and management of cardiac conditions [35, 36]. Moreover, abdominal imaging excels at evaluating organs like the liver and kidneys, contributing to the diagnosis of diverse pathologies. The inherent advantages of ultrasound, including portability and the absence of ionizing radiation, underscore its pivotal role in diagnostics, making it an invaluable asset in diverse clinical scenarios.

3.1.5 Nuclear medicine

Nuclear medicine, a dynamic field, revolves around the administration of radioactive tracers emitting gamma rays. In this process, detectors capture emitted gamma rays, and sophisticated computer processing translates this data into images depicting the tracer's distribution within the body. The core strength of nuclear medicine lies in functional imaging, which provides unique insights into organ function and metabolism. Myocardial perfusion imaging, a notable application, provides crucial information about heart function, aiding in the diagnosis and management of cardiovascular conditions [37, 38]. Similarly, PET plays a pivotal role in cancer staging, offering detailed information about tumor activity, and contributing significantly to the evaluation of neurological disorders. The ability of nuclear medicine to unravel physiological processes at a molecular level underscores its importance in personalized medicine, contributing to precise diagnostics and targeted therapeutic approaches.

3.1.6 Positron Emission Tomography (PET)

PET is an advanced imaging technique where a small amount of a radioactive substance, i.e. a glucose analog is introduced into the body. The emitted positrons produce gamma rays, whose detection enables the creation of functional images that vividly reflect metabolic activity. This process is particularly invaluable in oncology, where PET is extensively employed for cancer staging, treatment planning, and monitoring treatment response [39, 40]. Its applications extend to cardiology, facilitating the assessment of myocardial viability, and neurology, providing detailed insights into brain function.

3.1.7 Mammography

Mammography, an essential tool in breast imaging, harnesses X-rays to craft detailed images of breast tissue. The advent of digital mammography has augmented image quality while enhancing the efficiency of image storage and retrieval. Positioned as a primary method for breast cancer screening and diagnosis, mammography plays a crucial role in the early detection of breast cancer [41, 42]. Regular mammographic screenings are instrumental in improving treatment outcomes by facilitating early interventions and contributing to the comprehensive management of breast health.

3.1.8 Fluoroscopy

Fluoroscopy is a continuous X-ray imaging method that allows for the real-time monitoring of moving components within the body. Numerous interventional treatments use its adaptability, providing dynamic, real-time guidance. Conventional procedures include barium studies for assessing contrast agents' movement, catheter insertion, and joint injections [43, 44]. Fluoroscopy's real-time feedback is essential in interventional radiology and other medical fields, providing accuracy and precision in a wide range of operations, from diagnostics to treatments.

Overall, the variety of medical imaging modalities plays an important role in healthcare, providing professionals with an extensive range of tools for diagnosis and treatment. Each modality has distinct benefits, enabling an extensive knowledge of anatomical structures and clinical circumstances. Advancements in these technologies are continuously improving their capabilities, leading to better patient care, early illness identification, and progress in medical research [45]. Combining these modalities into clinical workflows emphasizes their importance in contemporary healthcare, influencing the diagnostic and treatment processes and eventually enhancing the results achieved for patients.

3.2 Significance of medical imaging in healthcare

Medical imaging has importance beyond collecting visual data, ranging beyond technical complexities. This section explores the important impact of medical imaging on the diagnostic and therapeutic components of healthcare. Medical imaging provides a crucial role in early illness detection and treatment planning, which makes it a key instrument for healthcare providers. The study continues throughout how imaging technologies help in precision medicine, monitoring treatment responses, and non-invasive assessments of both anatomy and function [46, 47]. The predominant theme throughout every application is the significant impact of medical imaging on patient outcomes, medical research, and the ongoing development of healthcare methods.

3.2.1 Early diagnosis and disease detection

One of the paramount contributions of medical imaging is its unparalleled ability to facilitate early diagnosis and the detection of diseases. Technologies such as X-ray, MRI, CT, and ultrasound empower clinicians to visualize internal structures, identify abnormalities, and detect diseases at their nascent stages [48, 49]. In the realm of oncology, early detection through modalities like mammography and PET-CT significantly enhances the chances of successful treatment, underscoring the critical role of medical imaging in the fight against cancer.

3.2.2 Treatment planning and precision medicine

Medical imaging serves as a linchpin in treatment planning, enabling clinicians to formulate precise and personalized intervention strategies. In orthopedics, imaging modalities like MRI and CT guide surgical planning for joint replacements or spinal surgeries. Similarly, in cardiology, imaging techniques such as echocardiography and coronary angiography aid in planning interventions like angioplasty or heart surgeries [50, 51]. The integration of imaging into treatment planning not only enhances precision but also minimizes the invasiveness of procedures, contributing to improved patient outcomes.

3.2.3 Monitoring disease progression and treatment response

The significance of medical imaging extends beyond diagnosis and initial treatment planning to the continuous monitoring of disease progression and treatment response. Sequential imaging studies, involving modalities like MRI or CT, provide clinicians with real-time insights into changes in tumor size, tissue healing, or the efficacy of therapeutic interventions [52, 53]. This real-time monitoring is crucial in oncology, neurology, and various other specialties, guiding adjustments in treatment protocols and ensuring timely interventions.

3.2.4 Non-invasive assessment of anatomy and function

Medical imaging facilitates a non-invasive assessment of both anatomical structures and physiological functions. Modalities like functional MRI (fMRI) and nuclear medicine techniques offer insights into brain activity, metabolism, and organ function. This non-invasiveness is particularly crucial in pediatrics and geriatrics, where traditional invasive diagnostic procedures may pose additional risks [54, 55]. Imaging's ability to unveil both structural and functional aspects of the body provides a comprehensive understanding, laying the foundation for informed clinical decision-making.

3.2.5 Image-guided interventions

The integration of medical imaging and interventional procedures has revolutionized medical practice. Techniques like fluoroscopy and ultrasound-guided interventions allow clinicians to perform procedures with unprecedented precision [56]. From guided biopsies in oncology to catheter placements in cardiology, image-guided interventions enhance accuracy, reduce complications, and offer minimally invasive alternatives to traditional surgical procedures.

3.2.6 Advancements in research and medical knowledge

Medical imaging contributes significantly to advancements in research and the expansion of medical knowledge. Imaging studies generate data that fuels research endeavors, enabling scientists and clinicians to explore disease mechanisms, evaluate treatment efficacy, and develop innovative diagnostic tools [57]. Medical imaging provides a wealth of information that deepens our understanding of disease processes, paving the way for groundbreaking discoveries and advancements in medical science.

3.2.7 Improved patient outcomes and quality of care

Ultimately, the overarching significance of medical imaging lies in its profound impact on patient outcomes and the overall quality of healthcare. Early diagnosis, precise treatment planning, and continuous monitoring facilitated by medical imaging contribute to improved survival rates, reduced morbidity, and enhanced patient well-being [58, 59]. The non-invasive nature of imaging procedures also aligns with patient-centric care, emphasizing safety and comfort.

3.2.8 Training and education

Medical imaging plays a vital role in medical education and training. From anatomy classes to specialized radiology training, medical imaging provides a tangible and visual representation of the human body. Interactive simulations and virtual reality applications based on imaging data enhance the educational experience for medical students and practitioners [60]. Understanding the nuances of medical imaging becomes an integral part of the skill set for healthcare professionals across various specialties.

3.2.9 Technological advancements and innovation

The relentless pursuit of excellence in medical imaging has led to continuous technological advancements and innovation. From the advent of digital imaging to the integration of artificial intelligence (AI) in image analysis, these innovations enhance the capabilities of imaging modalities [61, 62]. AI algorithms aid in rapid image interpretation, improving diagnostic accuracy and efficiency. Moreover, developments in imaging technologies, such as 3D imaging and functional imaging, open new frontiers in diagnostics and research.

3.2.10 Multimodal approaches for comprehensive assessment

The emergence of multimodal approaches, which integrate information from multiple imaging modalities for a comprehensive assessment, further underscores the significance of medical imaging [63]. Managing multimodal medical imaging data involves various issues and possibilities for reducing noise and enhancing images. Researchers may enhance image quality and diagnosis accuracy by combining data from several imaging modalities or modalities with varied noise characteristics to use their complementary strengths. Integrating methods like data fusion from MRI, CT scans, PET scans, and other imaging modalities enables an extensive assessment of anatomical structures and physiological processes. Improved signal processing techniques designed for multimodal data, such as multi-sensor fusion and joint estimation algorithms, allow for extracting more information while reducing noise and artifacts present in each modality. ML methods, such as DL structures, provide powerful ways to combine multimodal data and understand the connections between various imaging techniques and biological processes. Further study in this field is essential to fully using the capabilities of multimodal medical imaging, resulting in enhanced diagnostic precision, personalized medical approaches, and improved treatments for patients.

An overview of essential components in the domain of medical imaging is effectively and meticulously presented in Table 2. Multimodal integration, point-of-care imaging, precision oncology, interventional imaging, early disease monitoring, treatment guidance, functional neuroimaging, and non-invasive assessment are among the numerous contributions explored. Each contribution comprises a brief description, key metrics, modalities involved, and impact on healthcare. The data provided in Table 2 is very useful in realizing the diverse and nuanced ways in which medical imaging contributes to the enhancement of treatment accuracy, patient outcomes, and diagnostic capabilities. In conclusion, the significance of medical imaging in healthcare is profound and far-reaching. From the

Table 2 Comprehensive ov	erview of medical imaging contribu	itions		
Contribution	Modalities	Key Metrics	Impact on Healthcare	Description
Early Diagnosis	X-ray, MRI, CT, Ultrasound	Sensitivity, Specificity, False- Positive Rate	Improves treatment outcomes, reduces morbidity, and enhances the potential for successful interventions	Enables early detection of diseases and abnormalities, providing a crucial head start in treatment
Treatment Guidance	X-ray, MRI, CT	Accuracy, Precision, Navigational Accuracy	Enhances treatment precision, minimizes invasiveness, and contributes to personalized medicine approaches	Guides healthcare professionals in planning and executing precise treatment strategies
Disease Monitoring	MRI, CT	Temporal Resolution, Spatial Resolution, Monitoring Duration	Facilitates adaptive treatment strategies, aids in timely adjust- ments, and provides insights into overall patient progress	Enables real-time monitoring of illness progression, therapy effectiveness, and recovery after treatment
Non-Invasive Assessment	X-ray, MRI, CT, Ultrasound	Patient Comfort, Repeatability, Non-Ionizing Radiation	Enhances patient safety, allows for repeated assessments without significant risk, and contrib- utes to a more patient-friendly approach	Provides a non-invasive means of assessing anatomical structures and physiological functions
Multimodal Integration	PET-MRI, SPECT-CT, PET-CT	Fusion Accuracy, Information Coherence, Diagnostic Synergy	Enables a more holistic under- standing of complex medical conditions	Combines data obtained from vari- ous imaging techniques to provide an in-depth review of a patient's health
Point-of-Care Imaging	Portable ultrasound, handheld X-ray, mobile MRI	Speed, Accessibility, Real-Time Imaging Capabilities	Facilitates rapid decision-making and treatment initiation	Brings imaging capabilities directly to the patient's bedside for quick and efficient diagnostic assess- ments
Precision Oncology	PET, CT, MRI	Molecular Imaging, Tumor Het- erogeneity Assessment	Optimizes cancer treatment outcomes and minimizes side effects	Utilizes imaging to adapt treatments for cancer based to specific patient features and tumor biology
Interventional Imaging	Fluoroscopy, MRI-guided inter- ventions, CT angiography	Real-Time Imaging, Accuracy of Needle Guidance	Improves procedural success rates and patient outcomes	Guides minimally invasive pro- cedures, enhancing precision and reducing the invasiveness of treatments

Table 2 (continued)

 $\underline{\textcircled{O}}$ Springer

Contribution	Modalities	Key Metrics	Inpact on Healthcare	Description
runcuonal Neuroimagin	INIKI, FET	Connectivity Mapping, Task- Related Activation	Auvances understanding of prain function and aids in neurology	Focuses on maging oran tunc- tion and neurological disorders, providing insights into cognitive processes

early detection of diseases to precise treatment planning, continuous monitoring, and contributions to medical knowledge, imaging technologies have become indispensable tools in the modern healthcare landscape [64]. As technology continues to evolve, the role of medical imaging is poised to expand, promising further innovations and improvements in patient care. Medical imaging intricately weaves the journey from discovery to diagnosis, treatment, and beyond, shaping the future of healthcare and advancing the possibilities of personalized and effective medical interventions.

3.3 Datasets for medical images

Datasets are essential in medical image analysis research for training, validating, and testing ML and DL algorithms. Various publicly accessible datasets have been optimized for specific applications and imaging modalities, enabling a wide range of investigations in the medical imaging field. The MICCAI datasets provide resources for challenges like segmentation, registration, and classification, which include datasets like the Alzheimer's Disease Neuroimaging Initiative (ADNI) and the Multimodal Brain Tumor Image Segmentation (BRATS) challenge. TCIA provides cancer-related imaging data for tumor diagnosis and treatment response evaluation, while MSD provides datasets for segmentation tasks involving various anatomical structures and imaging modalities. IXI, LIDC-IDRI, and CAME-LYON represent important datasets that provide essential information for brain imaging, lung cancer screening, and histopathology analysis, respectively. The datasets provide support for several research projects in medical image processing, including illness diagnosis, treatment planning, and outcome prediction.

4 Noise removal in medical imaging

Medical imaging plays a pivotal role in modern healthcare, providing clinicians with valuable insights into the human body's structures and functions. However, the quality of medical images can be compromised by various sources of noise, presenting challenges to accurate diagnosis and treatment planning [65]. This section explores the intricate landscape of noise removal in medical imaging, spanning from the sources of noise to the comparative analysis of traditional and ML-based noise reduction techniques.

4.1 Sources of noise

Noise in medical images denotes to unwanted variations or distortions that can obscure true anatomical or physiological information. These variations can stem from multiple sources, including the image acquisition process and environmental factors.

4.1.1 Quantum noise

Quantum noise is inherent in medical imaging modalities that use X-ray photons, such as X-ray radiography and CT. It arises from the probabilistic nature of photon interactions with tissues during image acquisition [66]. As X-ray photons pass through the body, their interactions create an image based on the varying absorption characteristics of different

tissues. However, due to the discrete nature of photons, the resulting image exhibits fluctuations in pixel intensity, introducing uncertainty.

Mitigating quantum noise involves finding a balance between using a sufficiently high number of photons for image clarity and minimizing the radiation dose. Advanced techniques, such as statistical iterative reconstruction algorithms, aim to reduce quantum noise while maintaining diagnostic image quality.

4.1.2 Electronic noise

Electronic noise stems from imperfections in the electronic components of imaging devices, affecting the accuracy of signal detection. In digital imaging systems, electronic noise can manifest as random variations in pixel values, adding an additional layer of uncertainty to the acquired image [67]. This noise can be particularly pronounced in high-sensitivity imaging scenarios, impacting the detection of subtle features.

To address electronic noise, technological advancements focus on improving the signalto-noise ratio through enhanced electronic components and signal processing algorithms. Additionally, regular calibration and maintenance of imaging equipment are essential to minimize electronic noise and ensure accurate signal detection.

4.1.3 Radiation interference

External sources of radiation, such as cosmic rays, can interfere with the imaging process, introducing additional unwanted signals. This interference contributes to random spikes or fluctuations in pixel values, making it challenging to distinguish between radiation-induced noise and diagnostically relevant information [68].

We commonly employ shielding measures to reduce radiation interference. Implementing physical barriers and utilizing lead shielding can help protect imaging systems from external radiation sources [69]. Furthermore, during image processing, one can employ sophisticated algorithms to identify and filter out radiation-induced noise.

4.1.4 Temperature variations

Fluctuations in temperature can impact the performance of imaging equipment, influencing the generation and transmission of image signals. Temperature-related variations can introduce systematic errors in image acquisition, affecting the overall image quality [70]. In extreme cases, temperature variations can lead to thermal noise, which appears as random fluctuations in pixel values.

To minimize temperature-related noise, maintaining consistent imaging conditions through thermal stabilization methods is crucial [71]. Climate-controlled environments and cooling systems help regulate temperature, ensuring stable imaging conditions and reducing the impact of temperature-related noise on image quality.

4.1.5 Motion artifacts

Patient movement during image acquisition can result in motion artifacts, introducing distortions in the final image. Motion artifacts can lead to blurring or misalignment of anatomical structures, impacting diagnostic accuracy [72]. This source of noise is particularly relevant in modalities like MRI and PET, where motion can compromise image quality. Motion correction techniques play a pivotal role in mitigating artifacts associated with patient movement. Advanced imaging systems may utilize real-time tracking or retrospective motion correction algorithms to compensate for motion-induced distortions and ensure the accuracy of the final images [73].

4.1.6 Speckle noise

Imaging modalities that use ultrasound, such as ultrasonography, observe speckle noise, a unique type of noise. It appears as granular patterns in images, impacting image clarity and diagnostic interpretation. Interference patterns in the reflected ultrasound waves primarily cause speckle noise [74].

Ultrasound images employ several techniques to mitigate speckle noise. Image filtering methods, such as adaptive filtering and speckle reduction filters, aim to suppress speckle while preserving diagnostically relevant information [75]. Additionally, advanced ultrasound imaging systems may incorporate synthetic aperture imaging and advanced beamforming techniques to reduce speckle noise.

4.1.7 Systematic noise

Systematic noise refers to non-random noise that follows a specific pattern or distribution. It can result from systematic errors in the imaging system, affecting the overall uniformity of the image [76]. These errors may include sensor calibration inaccuracies, non-uniformities in detector response, or imperfections in imaging equipment.

Calibration procedures and quality assurance protocols are essential for minimizing systematic noise [77]. Regular checks and corrections for systematic errors contribute to maintaining image quality and ensuring accurate diagnostic information.

4.1.8 Patient anatomy variability

Variations in patient anatomy contribute to noise, especially in tasks such as image segmentation and comparison across different individuals [78]. Differences in organ shapes and sizes among patients introduce variability that can complicate image analysis.

We employ advanced image processing techniques, such as atlas-based segmentation and ML algorithms, to address patient anatomy variability [79]. These methods leverage large datasets to account for anatomical variations and enhance the robustness of image analysis algorithms.

4.1.9 Scanner artifacts

Imperfections or glitches in the imaging equipment can introduce artifacts into medical images. These artifacts may appear as streaks, lines, or distortions in the final image, impacting diagnostic accuracy [80]. Scanner artifacts can result from malfunctions in hardware components, software errors, or issues during the image acquisition process.

To mitigate scanner artifacts, regular maintenance and quality control checks are crucial. Additionally, advancements in imaging technology focus on improving hardware and software components to minimize the occurrence of artifacts.

4.1.10 Chemical noise

Chemical noise is particularly relevant in imaging modalities that involve chemical processes, such as MRI [81]. Changes in chemical make-up, like where contrast agents are distributed or the properties of the tissue, can cause changes in signal intensity, which can make it harder to tell the difference between tissues.

We design advanced MRI sequences and pulse sequences to minimize chemical noise. Additionally, the development of new contrast agents and imaging protocols aims to enhance the specificity of chemical information while reducing noise-related uncertainties.

4.1.11 Salt and pepper noise

Salt and pepper noise, also known as impulse noise, is a distinct form of digital image corruption characterized by the random occurrence of individual pixels with either extremely high or low intensity values, resembling grains of salt and pepper scattered throughout the image. This type of noise is introduced during the image acquisition process due to various factors, including sensor malfunctions, transmission errors, or external interference [82, 83]. The severity of salt and pepper noise is determined by the density of these extreme value pixels within the image. Its presence significantly degrades image quality, creating visually noticeable artifacts that can obscure critical details. Mitigation techniques involve the application of filters, such as median filtering, which replaces noisy pixels with the median value of neighboring pixels, effectively reducing the impact of salt and pepper noise while preserving essential image features. In applications like medical imaging, where precise interpretation is crucial, addressing salt and pepper noise is essential for maintaining the reliability of diagnostic processes.

4.1.12 Gaussian noise

Gaussian noise, a prevalent form of random signal disturbance, is characterized by pixel intensity variations following a Gaussian distribution. In digital images, each pixel's intensity is perturbed by a random value drawn from this distribution, resulting in a symmetrical bell-shaped curve centered on the mean value. The standard deviation parameter determines the spread of the distribution, influencing the degree of noise present. Originating from factors such as electronic or thermal fluctuations during image acquisition, Gaussian noise can adversely affect image quality by introducing undesirable artifacts like blurring and reduced contrast [84, 85]. Some ways to reduce the effects of Gaussian noise are to use smoothing filters, like the Gaussian filter, and advanced denoising algorithms that use statistical and ML techniques to keep the image's quality while reducing its effects. Understanding and managing Gaussian noise are crucial across various domains, from digital photography to medical imaging, to ensure accurate interpretation and reliable analysis of visual content.

The quality of diagnostic information heavily relies on the effective mitigation of various sources of noise. Table 3 outlines the key mitigation strategies employed to address different types of noise encountered in medical images. These strategies are crucial for maintaining image quality, minimizing artifacts, and ensuring the reliability of medical imaging systems. This comprehensive overview serves as a guide for healthcare professionals, image processing experts, and researchers navigating the intricate landscape of

Table 3 Noise mitigation strategies in medical imaging

Aspect	Calibration	Signal Pro- cessing	Shielding	Thermal Stabili- zation	Climate Control	Motion Correction Techniques	Gaussian Filters	Denoising Algo- rithms
Quantum Noise	>	×	×	×	×	×	×	×
Electronic Noise	>	>	×	×	×	×	>	>
Radiation Interference	×	×	>	×	×	×	>	>
Temperature Variations	×	×	×	>	>	×	×	×
Motion Artifacts	×	×	×	×	×	>	×	×
Speckle Noise	×	×	×	×	×	×	×	>
Systematic Noise	>	×	×	×	×	×	×	×
Patient Anatomy Variability	×	×	×	×	×	×	×	×
Scanner Artifacts	×	×	×	×	×	×	×	×
Chemical Noise	×	×	×	×	×	×	×	×
Salt and Pepper Noise	×	×	×	×	×	×	×	>
Gaussian Noise	×	×	×	×	×	×	>	>

noise mitigation in medical imaging. In the end, addressing the sources of noise in medical images is essential for maintaining image quality and improving diagnostic accuracy. Implementing mitigation strategies, including calibration, shielding, thermal stabilization, and motion correction, plays a crucial role in minimizing the impact of noise and optimizing the performance of medical imaging systems.

4.2 Traditional noise reduction techniques

Traditional noise reduction techniques in medical imaging have played a crucial role in addressing the inherent challenges posed by various sources of noise [86, 87]. Figure 1 illustrates traditional noise reduction techniques employed in the field of medical imaging, and the significance of these methods lies in their ability to enhance image quality by effectively minimizing unwanted variations while preserving diagnostically relevant information.

4.2.1 Spatial filtering

Spatial filtering is a fundamental technique in image processing that involves manipulating pixel values directly in the spatial domain, focusing on the local characteristics of the



Fig. 1 Traditional noise reduction techniques in medical imaging

image. Smoothing filters, a common category of spatial filters, are employed to reduce high-frequency noise by averaging pixel values within a local neighborhood [88, 89]. Two prominent examples of smoothing filters are the Gaussian filter and the median filter.

a) Gaussian Filter

The Gaussian Filter is a widely utilized smoothing filter known for its effectiveness in reducing high-frequency noise while preserving essential image details. It operates by convolving the image with a Gaussian kernel, which is a two-dimensional distribution resembling a bell-shaped curve [90, 91]. The convolution process involves assigning more weight to central pixels and gradually decreasing weights for pixels farther from the center. This weighted averaging blurs the image, effectively smoothing out variations caused by high-frequency noise. The parameter controlling the spread of the Gaussian distribution influences the degree of blurring, allowing for adaptability based on the specific noise characteristics and desired image quality. The Gaussian filter is particularly suitable for scenarios where noise manifests as random variations in pixel values, contributing to a more visually appealing and diagnostically relevant image.

b) Median Filter

The median filter is another spatial filtering technique designed to reduce noise, particularly impulsive noise such as salt and pepper noise. Instead of averaging pixel values, the Median Filter replaces each pixel's value with the median value within its local neighborhood. Because extreme values less affect the median than the mean, this approach effectively preserves edge details. In the context of impulsive noise, where individual pixels have exceptionally high or low intensity values, the median operation helps to mitigate the impact of these outliers [92, 93]. The Median Filter excels in scenarios where preserving sharp transitions and fine details is crucial, making it a valuable tool in medical imaging applications. It is particularly robust in situations where other smoothing filters might compromise critical image features.

4.2.2 Frequency domain filtering

Frequency domain filtering is a powerful technique in image processing that involves transforming the image into its frequency components using mathematical transformations. One of the cornerstone methods in frequency domain filtering is the Fourier transform, which represents the image as a sum of sinusoidal functions in the frequency domain [94, 95]. This method proves invaluable for noise reduction by manipulating specific frequencies associated with noise characteristics.

a) Fourier Transform

The Fourier transform is a mathematical operation that decomposes a signal, such as an image, into its frequency components. Image processing expresses the image as a sum of sinusoidal functions of varying frequencies. This transformation provides a unique representation of the image, revealing the frequency content that might be obscured in the spatial domain [96, 97]. Certain frequency ranges tend to concentrate noise, and we can achieve noise reduction by analyzing and manipulating these frequencies. The Fourier transform is particularly adept at identifying and isolating specific frequencies associated with noise patterns.

b) Low-pass Filtering

Low-pass filtering is a common approach in frequency domain filtering where highfrequency components associated with noise are suppressed, allowing only low-frequency components to pass through. This technique is effective when noise manifests as high-frequency variations in the image. By selectively attenuating the high-frequency noise components, low-pass filtering helps smooth out the image while preserving low-frequency details. This approach is analogous to spatial filtering with a Gaussian filter in the spatial domain [98, 99]. The controlled suppression of high-frequency noise ensures that the essential features of the image remain intact, contributing to improved image quality.

c) Band-pass Filtering

Another approach to frequency domain filtering is band-pass filtering, which selectively allows a certain range of frequencies to pass through while attenuating others. This technique is particularly useful when noise exhibits distinct frequency characteristics. In medical imaging, where different tissues and structures may contribute to specific frequency components, band-pass filtering allows for targeted noise reduction without compromising diagnostically relevant information [100, 101]. By customizing the passband to match the frequency range associated with noise, this method offers a nuanced and adaptive approach to noise reduction.

Frequency domain filtering provides a global perspective on noise reduction, impacting the entire image based on its frequency content. This approach is advantageous when noise exhibits specific frequency characteristics that are challenging to address in the spatial domain [102]. By unveiling the frequency composition of the image, frequency domain filtering allows for precise manipulation and suppression of noise components. Its versatility and adaptability make it a valuable tool in a variety of noise reduction applications, contributing to the continual improvement of image quality in medical imaging modalities.

4.2.3 Statistical methods

Statistical methods for noise reduction use mathematical transformations on image data to look at and change the image in a way that considers both its statistical properties and the noise's properties [103]. The Fourier transform, discussed earlier, is one such method. Another powerful technique in this category is the wavelet transform, which provides a nuanced approach to noise reduction by decomposing the image into different frequency bands.

a) Fourier Transform

The Fourier transform is a mathematical operation that represents an image in the frequency domain by decomposing it into sinusoidal functions of various frequencies. This transformation allows for the identification and isolation of specific frequencies associated with noise patterns. By manipulating these frequencies, noise reduction can be achieved [104, 105]. The Fourier transform provides a global perspective on the frequency composition of the image, enabling targeted suppression of noise components. Its statistical foundation lies in the systematic analysis of the image's frequency characteristics, contributing to effective noise reduction strategies.

b) Wavelet Transform

The wavelet transform is a statistical method that operates by decomposing the image into different frequency bands, known as wavelets. Unlike the Fourier transform, which represents the entire image in the frequency domain, the wavelet transform provides a multi-resolution analysis. This decomposition enables targeted noise reduction in specific frequency ranges, offering a more nuanced approach compared to global frequency domain filtering. Wavelet denoising has proven effective in preserving fine details while reducing noise, making it particularly valuable in scenarios where maintaining image fidelity is crucial [106, 107]. The statistical foundation of the wavelet Transform lies in its ability to capture and analyze the image's frequency content at different scales, allowing for adaptive noise reduction strategies.

c) Systematic Analysis and Manipulation

Both the Fourier transform and the wavelet transform provide a systematic way to analyze and manipulate image data in the context of noise reduction. These transformations serve as powerful tools for understanding the underlying statistical properties of the image and noise. By representing the image in different domains, these methods facilitate the identification of noise characteristics and the development of targeted noise reduction strategies [108, 109]. The systematic nature of these transformations ensures a structured and principled approach to noise reduction, contributing to the enhancement of image quality in medical imaging.

In summary, statistical methods in noise reduction, exemplified by the Fourier transform and the wavelet transform, offer sophisticated approaches to analyzing and manipulating image data. These methods, rooted in mathematical principles, provide insights into the statistical properties of both the image and the noise. The utilization of these transformations in noise reduction strategies underscores their significance in the continual refinement of image processing techniques in the field of medical imaging.

4.2.4 Probability-based noise reduction

In the realm of noise reduction, statistical models grounded in probability distributions play a pivotal role. Two prominent methods in this category, Maximum Likelihood Estimation (MLE) and Bayesian Methods, harness the power of probability-based approaches to separate signals from noise, particularly in scenarios with complex noise characteristics.

a) Maximum Likelihood Estimation (MLE)

Maximum Likelihood Estimation is a common statistical method used in noise reduction, where parameters of a probability distribution are estimated to maximize the likelihood of the observed data. In the context of noise reduction, MLE is employed to estimate noise parameters and discern signal from noise [110, 111]. The fundamental idea is to find the parameter values that make the observed data most probable under a given statistical model. MLE is particularly effective when dealing with noise characterized by specific probability distributions. By tailoring the model to the expected noise characteristics, MLE provides a robust framework for estimating parameters and enhancing noise reduction strategies. Its adaptability to different probability distributions makes it a versatile tool in the arsenal of statistical noise reduction techniques.

b) Bayesian methods

Bayesian methods introduce a probabilistic framework by incorporating prior knowledge about the image and the noise. This approach allows for the construction of a probabilistic model that considers both prior information and observed data. Bayesian methods provide a robust way to differentiate between signal and noise, especially in scenarios with complex noise characteristics. The Bayesian framework involves updating the probability distribution of the parameters based on both prior knowledge and new data, resulting in a posterior probability distribution [112, 113]. This posterior distribution encapsulates the updated knowledge about the parameters, reflecting the interplay between prior beliefs and observed evidence. Bayesian methods are particularly advantageous in situations where prior information about the image or noise is available, contributing to a more informed and adaptive noise reduction strategy.

c) Probability-based methods

Probability-based methods, encompassing MLE and Bayesian methods, offer a principled way to model and estimate noise in the context of image processing. By grounding noise reduction strategies in probabilistic frameworks, these methods provide a systematic and adaptable approach to various noise scenarios [114, 115]. The use of probability distributions allows for a nuanced understanding of the underlying statistical properties of both the image and the noise. In scenarios where precise modeling of noise characteristics is possible, probability-based methods excel, leading to more effective noise reduction strategies tailored to the specific challenges posed by the imaging modality and noise profile.

Finally, statistical models and probability-based noise reduction methods, such as MLE and Bayesian methods, represent sophisticated approaches to handling noise in medical imaging. These methods offer a principled way to model, estimate, and differentiate between signal and noise, contributing to the continual advancement of noise reduction techniques in the field of medical imaging. Their adaptability and robustness make them integral components in the pursuit of enhancing image quality and diagnostic accuracy.

4.2.5 Adaptive filtering techniques

In the quest for refined noise reduction strategies, adaptive filtering techniques emerge as dynamic solutions that adjust their parameters based on the characteristics of the local image region. Among these techniques, the Wiener filter stands out as an exemplary adaptive filter, leveraging statistical properties to minimize the mean-squared error between the estimated and true signal. This adaptive nature makes it particularly effective in scenarios with varying noise characteristics, providing a nuanced and targeted approach to noise suppression.

a) Wiener Filter

The Wiener filter is a notable example of an adaptive filter that operates by minimizing the mean-squared error between the estimated and true signal. This filter adapts its parameters based on the statistical properties of both the signal and the noise. The fundamental principle is to find the optimal filter that minimizes the expected value of the squared difference between the estimated signal and the true signal [116, 117]. The Wiener filter excels in scenarios where the statistical properties of the signal and noise are known or can be reliably estimated. By adapting its parameters based on these properties, the Wiener filter provides an effective means of reducing noise while preserving the essential characteristics of the signal. Its adaptability to varying noise conditions makes it a versatile tool in situations where noise profiles may change across different regions of the image.

b) Adaptive Filtering Dynamics

Adaptive filtering provides a dynamic method for reducing noise. Adaptive filters modify their parameters according on various characteristics of the image, unlike fixed filters. This flexibility enables more precise and advanced noise reduction. Adaptive filtering is essential in medical imaging because of the diverse features of images [118, 119]. Optimizing the filter response to the distinct noise factors in various areas of an image enhances image quality and diagnostic precision.

c) Local Image Properties

The key to the effectiveness of adaptive filtering lies in its consideration of local image properties. Instead of applying a uniform filter across the entire image, adaptive filters analyze the characteristics of the local neighborhood around each pixel [120, 121]. This analysis enables the filter to dynamically adjust its parameters, responding to the variations in noise characteristics and signal intensity within different regions. As a result, adaptive filtering can effectively mitigate noise without compromising the crucial details of the underlying signal.

4.2.6 Deep learning algorithms with traditional noise

The combination of DL techniques with traditional noise reduction methods provides an approach to the issues of noise reduction in medical imaging and other fields. Traditional methods like spatial filtering, frequency domain filtering, and adaptive filtering are effective for noise reduction in images while maintaining visual features. But these techniques have difficulties adapting to challenging noise patterns and fluctuations across various types of medical images. DL algorithms improve at learning involved patterns and features from enormous datasets, making them suitable for employment opportunities involving noise reduction. Researchers can utilize the robustness and interpretability of traditional approaches together with the flexibility and scalability of DL by integrating both methodologies. Utilizing a hybrid approach enables DL models to benefit from enriched and preprocessed data produced by conventional noise reduction methods, leading to greater noise reduction capabilities and more precise medical picture interpretation. Furthermore, integrating DL with conventional methods provides opportunities for modifying and enhancing, resulting in personalized solutions designed for imaging modalities and noise features. The combination of DL algorithms with traditional noise reduction approaches shows a great deal of for improving the quality and dependability of medical imaging applications.

Table 4 Lit	erature review for noise reduction in medical imaging		
References	Contributions to Noise Reduction	Objectives	Results
[122]	Explored noise and uncertainty in medical imaging	Investigated methods for uncertainty quantification	Proposed techniques for improved noise characteriza- tion
[123]	Specialized in noise reduction methods for MRI	To develop and evaluate noise reduction strategies specific to MRI	Introduced advanced filtering techniques tailored for MRI noise
[124]	Comprehensive overview of medical imaging prin- ciples, including noise reduction	Provide a fundamental understanding of noise in medical images	Outlined general principles for effective noise reduc- tion
[125]	Explored various techniques in medical image analysis, including noise reduction	To review and compare different noise reduction methods	Comparative analysis of noise reduction techniques in medical images
[126]	Addressed image processing techniques in medical imaging, including noise reduction	Investigate the application of image processing methods for noise reduction	Presented a review of image processing techniques with a focus on noise reduction in medical images
[127]	Focused on imaging modalities using ultrasound, addressing challenges related to noise	Develop noise reduction strategies for ultrasound imaging	Proposed and evaluated novel approaches for reduc- ing noise in ultrasound images
[128]	Covered aspects of image processing in medical imaging, including noise reduction	Examine the role of image processing in addressing noise challenges	Provided insights into state-of-the-art noise reduction techniques in medical imaging
[129]	Explored digital image processing, including its application to noise reduction in medical images	Investigate digital processing techniques for noise reduction	Reviewed the efficacy of various digital processing methods for reducing noise in medical images

The objectives of the researchers and the results that were attained are effectively described in Table 4, which also provides an overview of noise reduction in medical imaging. Adaptive filtering techniques, demonstrated by the Wiener Filter, comprise a dynamic and improved methodology for mitigating noise in the field of medical imaging. The filters' ability to adapt to the unique features of the image allows for enhanced noise suppression. Adaptive filtering comes into its own as a valuable instrument for improving image quality and facilitating precise diagnosis in the dynamic field of medical image processing, where noise conditions can vary significantly.

4.3 Machine learning-based noise removal

The integration of ML methods has significantly transformed the domain of noise reduction in medical imaging in recent times. Traditional methods, while effective, usually rely on predetermined algorithms that might encounter difficulties when attempting to accommodate the complexities of various noise patterns. By permitting the algorithm to learn and adapt from the data itself, ML represents an innovation. This segment provides an overview of the core principles underlying ML-based noise reduction. It investigates how these methodologies utilize sophisticated algorithms to improve the precision and effectiveness of noise elimination in medical images. Also, the comprehensive understanding and innovative strategies employed by the authors to contribute to the field of noise reduction, specifically through the implementation of ML techniques, are emphasized in Table 5. The utilization of ML-based noise elimination methods in the field of medical imaging is illustrated in Fig. 2. These techniques have significantly improved the quality of images in a wide range of domains and applications.

ML is an area of artificial intelligence that enables computers to analyze data, identify patterns, and make predictions and assessments without being explicitly programmed. ML algorithms for noise reduction are trained on datasets that include both noisy and clean medical images. The algorithm learns the patterns and correlations between noisy input and clean output during training, enabling it to make accurate predictions on novel information. These methods use advanced algorithms to analyze data, adapt to various noise patterns, and improve image sharpness. Let's analyze numerous important ML-based noise reduction techniques in the field:

4.3.1 Neural network-based denoising

Neural networks, particularly DL architectures, have demonstrated remarkable capabilities in noise removal. CNNs excel in learning hierarchical features from medical images, allowing them to effectively discern between noise and genuine signal. Trained on large datasets, these networks can generalize well to diverse noise patterns, providing robust noise reduction across various imaging modalities [143].

4.3.2 Autoencoders for image restoration

Autoencoders, a type of neural network architecture, are adept at learning efficient representations of input data. In the context of noise removal, autoencoders can be trained on clean images to learn the underlying structure, and then applied to noisy images for restoration. Variational autoencoders introduce a probabilistic framework, allowing for more nuanced modeling of noise distributions and facilitating adaptive noise removal [144].

Table 5 Lit	erature review for machine learning-based noise remov	val in medical imaging	
References	Contributions to Noise Reduction	Objectives	Results
[130]	Pioneered Generative Adversarial Networks (GANs) for image denoising	Develop DL models for effective noise removal	Introduced GANs as a powerful tool for generating clean images from noisy inputs
[131]	Applied DL techniques for noise reduction in medi- cal images	Investigate the use of CNNs for noise reduction	Demonstrated the effectiveness of CNNs in reducing noise and preserving important features in medical images
[132]	Developed normalization techniques (Batch Nor- malization) to enhance noise reduction models	Improve the stability and performance of DL models for noise removal	Batch Normalization contributed to more stable and efficient training of deep neural networks for noise reduction
[133]	Proposed non-local means filtering using DL for image denoising	Explore non-local approaches within a DL frame- work	Introduced a novel DL-based non-local means filter, improving denoising performance compared to traditional methods
[134]	Explored deep CNNs for low-level vision tasks, including noise reduction	Investigate the potential of deep CNNs in handling noise at pixel-level	Provided insights into the capability of deep CNNs for noise reduction, paving the way for further advancements in low-level vision tasks
[135]	Applied ML to image inpainting, contributing to noise removal in damaged or missing regions	Develop algorithms for inpainting damaged areas in images	Presented techniques for utilizing ML in image inpainting, aiding in noise removal in specific regions of interest
[136]	Explored the integration of DL models with traditional image processing techniques for noise reduction	Combine the strengths of DL and classical methods	Demonstrated improved noise reduction by integrat- ing DL approaches with traditional image process- ing methods
[137]	Investigated the use of unsupervised learning for noise reduction in various imaging scenarios	Explore unsupervised learning techniques for adaptability to diverse noise patterns	Showcased the potential of unsupervised learning in adapting to and effectively reducing noise in differ- ent image types and environments
[138]	Optimizing Cancer Treatment using Artificial Neural Networks	Develop a method for optimal cancer treatment using artificial neural networks	Achieved improved accuracy and efficiency in cancer treatment optimization
[139]	Comparing the eruption pattern of second molars in skeletal class I and class III malocclusions in 8–9-year-old children	Investigate second molar eruption patterns in differ- ent malocclusions	Identified significant differences in eruption patterns between skeletal class I and class III malocclusions
[140]	Low-cost stochastic computing-based fuzzy filter for reducing image noise	Develop a low-cost fuzzy filtering method for image noise reduction using stochastic computing	Achieved significant reduction in hardware area and power consumption while preserving image quality

.; .: : ÷ u

Table 5 (co	ontinued)		
References	Contributions to Noise Reduction	Objectives	Results
[141]	Classification of FNIRS Signals using Ensemble Learning and Adaptive Neuro-Fuzzy Inference System	Classify FNIRS signals using ensemble learning and adaptive neuro-fuzzy inference system	Improved FNIRS signal classification accuracy and reduced standard deviation
[142]	Utilizing ML for detecting electrical faults in a large power system network	Utilize ML for monitoring electrical disturbances in wide power networks	Achieved improved accuracy in identifying disturbances and enhancing power system monitoring



Fig. 2 Machine learning-based noise removal for medical imaging

4.3.3 Generative Adversarial Networks (GANs)

GANs, that involve a generator and a discriminator in a competitive learning setup, have been utilized for noise reduction. The generator is taught to produce images with reduced noise, while the discriminator is trained to differentiate between images that are clean and those that are noisy. Adversarial training produces the development of genuine and attractive denoised images, highlighting the potential of GANs for noise reduction [145].

4.3.4 Non-local means filtering with machine learning

Non-Local Means (NLM) filtering, a classical method in image denoising, has been enhanced with ML. By incorporating ML models into the NLM framework, these methods adaptively adjust filter parameters based on the local image content. This fusion of traditional filtering techniques with ML intelligence enhances the overall denoising performance [146].

4.3.5 Image transformation networks

Image Transformation Networks use DL architectures for transforming noisy images into their prominent version. These networks obtain the mapping function during training and can ultimately be used for removing noise from input images. These networks provide efficient noise reduction via comprehending the interactions between noisy and clean image sequences [147].

ML-based noise reduction algorithms are an important development in medical imaging. These advanced algorithms, such as neural networks and GANs, use data-driven learning to improve picture quality, proving to be effective tools for producing better and diagnostically relevant medical images.

4.4 Comparative analysis

In the rapidly evolving field of medical image processing, the effectiveness of noise reduction methods is crucial in evaluating the quality of diagnostic information. This section compares conventional approaches with ML-based methods to analyze their strengths, errors, and overall performance.

4.4.1 Traditional noise removal techniques

Traditional noise removal techniques, rooted in well-established principles, have been the stalwarts of image processing for decades. Methods like spatial domain filtering, frequency domain filtering, and statistical approaches such as Fourier transformations and median filtering have proven efficacy in certain scenarios. However, these techniques rely on assumptions about noise characteristics and may struggle with adaptability to diverse noise profiles. Their performance can vary depending on the imaging modality, noise distribution, and the presence of complex noise patterns [148].

4.4.2 Machine learning-based noise removal

ML noise reduction methods represent an important development in image processing. Algorithms like neural networks, autoencoders, and GANs use data-driven learning. They demonstrate the capacity to adapt to various noise patterns via comprehending complex connections between clear and noisy images. Their versatility enables them to generalize well across many imaging modalities and noise conditions, enabling a more versatile and effective method for noise reduction [149, 150].

4.4.3 Deep learning algorithms with traditional noise reduction techniques

Integrating DL algorithms with traditional noise reduction methods has enormous possibilities for improving medical image analysis. CNNs proved their outstanding ability in learning nuanced patterns and features from extensive datasets, highlighting the strength of DL models. Combined benefits may be achieved by integrating DL with traditional noise reduction techniques like spatial filtering or wavelet denoising. DL models can effectively denoise medical imaging by using the comprehensive data included in pairs of noisy and noise-free images. This integration enables DL algorithms to effectively identify intricate structures and features while reducing noise artifacts, resulting in enhanced picture quality and diagnostic precision [151, 152]. By combining standard noise reduction techniques' computational economic growth with DL models' representational capability, this strategy can achieve better noise reduction performance than individual approaches. The development of integrated techniques has tremendous promise to advance medical image processing and improve healthcare diagnostics.

4.4.4 Comparative evaluation

- a) Adaptability: Traditional techniques may have difficulties in dealing with various and intricate noise patterns because to their reliance on predetermined assumptions about noise properties. ML-based approaches improve in flexibility, by learning from data and modifying parameters to deal with various noise patterns [153].
- b) Performance across Modalities: Traditional methods may perform well in specific scenarios but might falter when applied to different imaging modalities with distinct noise characteristics. ML-based approaches, trained on diverse datasets, highlight robust performance across various modalities, making them more versatile in real-world applications [154].
- c) Learning and Generalization: ML-based methods have the advantage of learning intricate patterns from large datasets, allowing them to generalize well to unseen data. Traditional methods, lacking the learning capacity of intelligent algorithms, might struggle to adapt to unforeseen noise scenarios [155].
- d) Computational Complexity: Conventional methods can be computationally efficient, yielding rapid results in real-time applications. ML methods, particularly DL models, may need increased computer resources for both training and inference [156]. Advancements in technology and optimization methods are reducing this concern.

A comparison between traditional and ML-based noise removal techniques is provided in Table 6. The outcomes indicate that although traditional techniques have a track record of dependability in specific situations, ML-based approaches are formidable competitors in the quest for superior image quality due to their adaptability, learning capability, and versatility. The specific demands of the imaging task, the characteristics of the noise, and computational efficiency factors will determine which of these methods is selected. The assets and limitations of both traditional and ML-based approaches are comprehensively illustrated in Fig. 3, which presents a comparative analysis of each criterion. The visual representations facilitate comprehension of the minor differences between these techniques by depicting the trends identified during the comparison. The field of medical imaging noise removal is increasingly transitioning towards intelligent, data-driven solutions, as technological progress, and the availability of diverse and extensive datasets for training ML models advance.

5 Image segmentation in medical imaging

Within medical image analysis, segmentation is an essential process that plays an important role in recognizing the spatial details of anatomical structures. Segmentation is important for improving accuracy, diagnosis, and treatment planning beyond mere visualization. Segmentation allows doctors to accurately identify anatomical structures, helping in targeted therapies and reducing unintended harm by navigating complicated anatomical landscapes.

Table 6 Comparative evaluation	1 of noise removal in medical imaging	
Comparative Evaluation	Traditional Techniques	ML-Based Approaches
Adaptability	Issues in controlling various and intricate noise patterns as the con- sequence of presumptions	Effective of adjusting parameters and learning from data to accommo- date an extensive variety of noise characteristics
Performance across Modalities	Extensively effective in specific circumstances but lacks consistency across modalities	Establishes resilient performance across a range of modalities, based on training on a diverse array of datasets
Learning and Generalization	Lack of ability for learning has difficulty generalizing to unseen data	Large datasets are utilized to identify intricate patterns, which general- ize well for novel scenarios
Computational Complexity	Frequently computationally efficient, producing immediate conse- quences	Highly expensive computationally, particularly when DL models are utilized

50
agin
<u> </u>
medica
н.
removal
noise
\mathbf{of}
evaluation
Comparative
5
Ť



Fig. 3 Comparative analysis of traditional and ML-based noise removal techniques

It provides the basis for treatment planning, enabling the adaptation of methods according to the unique features of each patient. Segmentation also plays an important role in quantitative analysis by providing important metrics to comprehend illness development and therapy response [157]. Segmentation helps in combining data from several imaging techniques to enhance the overall comprehension of the patient's condition. Advancements in artificial intelligence and ML have made segmentation more complex, providing automatic and precise analysis. Segmentation not only benefits clinical applications but also improves communication between healthcare teams and assists in training professionals and patients. Advancements in technology are increasing the importance of segmentation in medical image analysis, paving the way for imaging to not only diagnose but also strategically enhance patient care. The contributions of notable papers to image segmentation in medical imaging are presented in Table 7. The segmentation methods used include conventional convolutional networks as well as advanced models such as U-Net, variational autoencoders, and conditional generative adversarial networks. The goals of these efforts include accurate organ delineation, enhanced tumor identification, probabilistic segmentation, and instance-level segmentation for accurate lesion localization.

5.1 Segmentation using digital pathology

Digital pathology is an important field in medical imaging that provides high-resolution digital images of tissue samples for diagnostic and research applications. Digital pathology is crucial in image segmentation for recognizing the boundaries of cells, nuclei, or abnormal characteristics inside tissue samples. Digital pathology segmentation enables precise analysis, tumor identification, classification, and prediction, assisting pathologists in making precise diagnoses and treatment possibilities [158]. Image segmentation techniques such as thresholding, edge detection, or DL-based algorithms can separate digital pathology images into relevant sections for subsequent investigation. Integrating digital pathology with image segmentation methods improves the efficiency and accuracy of pathological analysis, resulting in improved patient care outcomes. Below is a literature review based on digital pathology:

Table 7 Lit	erature review for image segmentation in medical imag	ging	
References	Segmentation Techniques	Objectives	Results
[172]	U-Net architecture for organ segmentation in MRI	Accurate delineation of organs for treatment plan- ning	Achieved high segmentation accuracy with U-Net, enabling precise organ delineation in MRI scans
[173]	Multi-scale convolutional networks for tumor segmentation	Improve tumor detection and segmentation in medi- cal images	Demonstrated enhanced performance in tumor seg- mentation using multi-scale convolutional networks
[174]	Variational autoencoders for brain tissue segmenta- tion	Explore generative models for probabilistic seg- mentation	Introduced probabilistic segmentation using vari- ational autoencoders, providing uncertainty esti- mates in brain tissue segmentation
[175]	3D U-Net for volumetric segmentation of medical images	Extend U-Net for 3D segmentation tasks	Successfully applied 3D U-Net for volumetric seg- mentation, particularly useful in three-dimensional medical imaging datasets
[176]	Integration of attention mechanisms in CNNs for organ segmentation	Enhance CNNs with attention mechanisms for improved segmentation	Improved organ segmentation accuracy by incorpo- rating attention mechanisms into CNNs
[177]	Mask R-CNN for instance-level segmentation of lesions	Achieve instance-level segmentation for precise lesion localization	Implemented Mask R-CNN, enabling accurate instance-level segmentation of lesions in medical images
[178]	Conditional GANs (cGANs) for cardiac image segmentation	Generate realistic cardiac segmentation masks using cGANs	Successfully applied cGANs to generate realistic cardiac segmentation masks, improving accuracy in cardiac image segmentation
[179]	Graph-based methods for anatomical structure segmentation	Utilize graph-based approaches for capturing ana- tomical relationships	Demonstrated the effectiveness of graph-based methods in capturing anatomical structures and improving segmentation accuracy
[180]	Ensemble learning for multi-modal medical image segmentation	Combine information from different imaging modalities for improved segmentation	Showcased the benefits of ensemble learning in com- bining multi-modal information for more robust and accurate medical image segmentation
[181]	Deep CNNs for automated liver segmentation	Develop automated methods for liver segmentation in medical images	Proposed deep CNNs achieving high accuracy in automated liver segmentation, crucial for various liver-related medical applications
[182]	Transformer-based networks for brain tumor segmentation	Apply transformer architectures to medical image segmentation tasks	Demonstrated the effectiveness of transformer-based networks in capturing long-range dependencies for improved brain tumor segmentation

Table 7 (co	ontinued)		
References	Segmentation Techniques	Objectives	Results
[183]	Image registration and segmentation fusion for improved tumor delineation	Integrate registration and segmentation for more accurate tumor localization	Demonstrated improved tumor delineation by combining image registration and segmentation methods, enhancing localization accuracy
[184]	Capsule networks for hierarchical feature extraction in lung segmentation	Explore capsule networks for capturing hierarchical features in lung images	Introduced capsule networks to enhance lung segmentation by capturing hierarchical features, leading to improved performance
[185]	Transfer learning for rare disease segmentation using limited annotated data	Address segmentation challenges in rare diseases with limited labeled data	Applied transfer learning techniques to achieve robust segmentation performance in the context of rare diseases with limited annotated data
[186]	Dynamic contour-based segmentation for real-time cardiac imaging	Develop real-time segmentation methods for dynamic cardiac imaging	Proposed dynamic contour-based segmentation for real-time cardiac imaging, enabling precise seg- mentation in dynamic and fast-paced scenarios
[75]	Semi-supervised learning for tumor segmentation in histopathological images	Explore semi-supervised approaches for tumor segmentation in pathology images	Demonstrated the effectiveness of semi-supervised learning in histopathological image segmentation, leveraging both labeled and unlabeled data for improved results

Sasmal et al. provide a method to segregate epithelial layers in oral histopathology images. They use superpixel-based clustering together with an enhanced Nature-Inspired Optimization Algorithms (NIOAs) termed as the Cooperative Search (CS) algorithm. The CS algorithm integrates the Aquila Optimizer (AO) and Particle Swarm Optimizer (PSO) to improve exploration–exploitation capabilities and avoid being stuck in local optima. The paper compares three superpixel techniques with CS for the best epithelial layer segmentation. The results demonstrate that the suggested strategy outperforms current techniques, reaching high accuracy, MCC, Dice, and Jaccard scores. Furthermore, CS shows competitive optimization efficiency when tested on benchmark functions. This work introduces a new method in medical picture segmentation that combines superpixel methods with an enhanced optimization algorithm to provide more precise and efficient epithelial layer segmentation [159].

Dhal et al. analyze the difficulties related to the Fuzzy C-Means (FCM) method in image segmentation and provide remedies to tackle these problems. FCM, an effective clustering method, has limitations including high computing complexity, reliance on initial cluster centers, dependency on membership matrices, and susceptibility to noise [160]. The study provides an ongoing review of possibilities described in recent literature to address these obstacles. It also addresses the primary challenges for developing enhanced FCM versions. This work enhances the area of digital image processing by recognizing and overcoming the constraints of FCM in image segmentation.

Dhal et al. highlight the difficulties in pathology image segmentation due to variations in illumination and staining while collecting microscopic images. An Improved Slime Mould Algorithm (ISMA) is suggested, using opposition-based learning and a differential evolution mutation technique for illumination-free White Blood Cell (WBC) segmentation. The work thoroughly analyzes color components from different color spaces for clustering, showing the efficiency of illumination-independent and color component-focused methods for picture segmentation. The ISMA-KM algorithm combined with the "ab" bands of the CIELab color space are most effective for segmenting nuclei, while ISMA-KM with the "CbCr" color component of the YCbCr color space is most accurate for segmenting whole white blood cells. The ISMA method demonstrates comparable performance with existing NIOAs on CEC2019 benchmark test functions, indicating its potential for effective image segmentation [161].

Dhal et al. highlight the issues of local optima trapping and extended computing time linked to crisp partitional clustering methods such as K-Means (KM) in image segmentation. The authors suggest a crisp clustering approach called Chaotic Fitness-Dependent Quasi-Reflected Aquila Optimizer (CFDQRAO), which is an enhanced version of AO. It integrates chaotic fitness-dependent quasi-reflection-based Opposition Based Learning (OBL) to boost optimization efficiency. The research also investigates the use of Simple Linear Iterative Clustering (SLIC)-based superpixel images to decrease computing time. The CFDQRAO approach outperforms other Nature-Inspired Optimization based on data collected from blood pathology images. The SLIC-CFDQRAO clustering approach surpasses previous SLIC-NIOA and SLIC-KM algorithms in visual analysis and segmentation criteria for quality [162].

Sasmal et al. published an extensive review on combining superpixel images with clustering approaches for a variety of image segmentation objectives. The research highlights the need of choosing suitable superpixel generation methods and clustering algorithms to get precise and effective segmentation outcomes. The paper addresses the latest developments in superpixel synthesis and clustering techniques, emphasizing

their benefits and difficulties. The authors execute a comparative evaluation of superpixel-based clustering algorithms using oral pathology and leaf pictures to assess their effectiveness. Experimental findings show that superpixel-based clustering algorithms outperform standard clustering methods in terms of segmentation accuracy and quality metrics. The paper provides useful insights for researchers in the area and highlights potential areas of study such as automated superpixel production, integration of DL, and improving clustering effectiveness for noisy images [163].

Ray et al. focus on the automatic segmentation of epithelial layers in pathology images, crucial for disease detection. Employing PSO and KM clustering in the CIElab color space, the study aims to enhance Computer-Aided Diagnosis (CAD) systems. Comparative analysis using different color spaces highlights CIElab's superior performance. Experimental findings demonstrate PSO with CIElab achieving an impressive 98.43% accuracy, surpassing other methods. This research contributes to more accurate and efficient epithelial layer segmentation, advancing CAD-based disease detection systems [164].

Dhal et al. suggest a histogram-based fast fuzzy image clustering (HBFFIC) approach to tackle the obstacles of FCM clustering in image segmentation. This technique utilizes morphological reconstruction (MR) to improve resistance to noise while preserving image intricacies. Utilizing gray-level histograms for clustering leads to a significant decrease in processing time. The research addresses local optima concerns by using NIOAs, including the Archimedes optimizer (AO). The HBFFIC-AO hybrid algorithm improves state-of-the-art algorithms in segmenting synthetic and real-world pathology images, as shown by experimental data. This study enhances image segmentation algorithms for environments with noise [165].

Dhal et al. resolve difficulties with image segmentation by proposing a Histogram-Based Fuzzy Clustering (HBFC) method which includes an improved Firefly Algorithm (FA). FCM is a prominent clustering method that commonly faces issues with computational complexity and vulnerability to noise. The proposed Hybrid Bat Firefly Algorithm combines Firefly Algorithm with rough set-based population, random attraction, and local search methods. Clustering is conducted using gray-level histograms to minimize pixel misclassification. Comparison with state-of-the-art NIOAs and conventional methodologies shows that HBFC exceeds in accuracy, resilience, and segmented output quality. This study enhances fuzzy picture clustering methods to enhance the outcome of segmentation [166].

Iqbal et al. (2023) provide AMIAC, an Adaptive Medical Image Analysis and Classification framework that emphasizes adaptive self-learning for DL models in medical imaging. The framework tackles issues caused by changes in image distribution by using transfer learning, adaptive learning, and incremental learning methods. AMIAC enhances accuracy and efficiency by combining manual and auto CNN-based characteristics, thereby minimizing the need for manual retraining. The framework integrates manual characteristics with pretrained CNN models to improve performance in tasks such as tumor classification and lesion identification. The experimental findings show a high F1-score and precision, indicating the potential of AMIAC as a tool to aid pathologists [167].

Das et al. present a Histogram-based Fast and Robust Crisp Image Clustering (HFRCIC) technique for image segmentation, addressing issues with conventional K-means clustering. The technique incorporates morphological reconstruction for noise immunity and preservation of image details, enhancing clustering robustness. By clustering based on gray levels rather than individual pixels, computational time is reduced. To overcome local optima challenges, Stochastic Fractal Search (SFS) is employed for optimal cluster center

determination. Experimental results demonstrate the superiority of HFRCIC-SFS over existing segmentation algorithms and NIOA-based clustering techniques [168].

Ray et al. introduces the Whale Optimization Algorithm (WOA) for breast histopathology image segmentation, addressing challenges posed by correlated and noisy regions. The study focuses on automatic cancer cell detection using clustering techniques, emphasizing the sensitivity of traditional methods to initial cluster centers. Through a comparative analysis, the proposed approach demonstrates superior precision, robustness, and segmentation quality compared to existing clustering methods and nature-inspired optimization algorithms [169].

Ray et al. investigate superpixel-based methods for segmenting images, with a specific emphasis on medical imaging, notably kidney renal cell carcinoma images. The SLIC technique is used for its computational economy and high performance on pathology images. The study suggests that using SLIC with PSO outperforms other approaches in segmentation accuracy when compared to ground truth images, by using PSO and KM clustering techniques with superpixel preprocessing [170].

Dhal et al. suggest an innovative hybridization method that merges the Sine–Cosine Algorithm (SC) with KM for categorizing pathology images. Their study focuses on improving visual information extraction and grouping for cancer studies by using the NIOAs and ML approaches to better nuclei segmentation. The hybrid SC-KM method was created to overcome the constraints of SC and KM algorithms, providing better outcomes than conventional clustering models like K-Means, GA, PSO, and the SC algorithm [171].

This section explores the pivotal role of segmentation in medical image analysis, delves into the landscape of traditional segmentation methods, introduces the transformative realm of ML-based segmentation, and concludes with a comparative analysis, shedding light on the strengths and limitations of each approach (Fig. 4).



5.2 Traditional segmentation methods

In the realm of medical image analysis, traditional segmentation methods stand as the bedrock, providing clinicians with foundational tools to delineate and isolate anatomical structures for precise diagnosis and treatment planning. This section offers an insightful exploration into classical image segmentation techniques, elucidating their overarching principles and applications.

5.2.1 Contour-based methods

Traditional image segmentation frequently utilizes contour-based techniques to establish boundaries of structures for segmentation. Edge detection techniques and active contours use gradient information and energy reduction to define object boundaries [187]. Although efficient in situations with well-defined boundaries, these techniques may encounter difficulties with shapes that are irregular and fluctuations in intensity.

5.2.2 Thresholding techniques

Thresholding is a simple yet powerful segmentation approach that involves separating regions based on intensity levels. By setting a threshold value, pixel intensities below or above the threshold are classified into distinct regions [188]. This method is particularly effective in scenarios where there is a clear contrast between the object of interest and the background. However, it might be sensitive to noise and variations in intensity.

5.2.3 Region-growing algorithms

Region-growing algorithms start with seed points and iteratively add neighboring pixels that meet certain criteria, forming cohesive regions [189]. This approach is suitable for images with homogenous regions and gradual intensity transitions. However, its performance can be influenced by the choice of seed points and is sensitive to noise.

5.2.4 Clustering methods

Clustering techniques, such as k-means clustering, group pixels with similar intensity values into clusters. This segmentation method is particularly effective in scenarios with distinct intensity distributions [190]. However, the accuracy of clustering-based segmentation heavily relies on the appropriate choice of the number of clusters and initial cluster centers.

5.2.5 Watershed transform

The watershed transform is inspired by the concept of flooding a landscape and marking regions where flooding converges. It is particularly useful for segmenting images with

objects having different intensities [191]. However, the watershed transform might oversegment images with fine details, and careful pre-processing is regularly required.

5.2.6 Deformable models

Deformable models, such as snakes or active contours, are mathematical models that evolve to fit object boundaries in the image [192]. They are advantageous in capturing intricate shapes and contours. However, their performance can be influenced by initialization and may struggle with concavities and irregularities.

In summary, classical image segmentation techniques provide a diverse toolkit for delineating structures in medical images. While each method has its strengths, the choice depends on the characteristics of the image, the nature of the structures of interest, and considerations of computational efficiency. As the foundation of segmentation methodologies, traditional approaches pave the way for a deeper understanding of anatomical land-scapes in medical image analysis.

5.3 Machine learning-based segmentation

In the dynamic landscape of medical image analysis, the advent of ML and DL has ushered in a change in basic assumptions, revolutionizing the field of segmentation. This section provides a comprehensive introduction to ML and DL methods tailored for segmentation tasks, elucidating the transformative capabilities that these intelligent algorithms bring to the precision and efficiency of delineating anatomical structures in medical images.

5.3.1 Machine learning primer

ML, at its core, involves the development of algorithms that enable computers to learn patterns and make predictions or decisions without explicit programming. In medical image segmentation, ML leverages training datasets to learn relationships between input images and segmented structures [193]. Algorithms like Support Vector Machines (SVM), Random Forests, and Decision Trees have been employed, each with its strengths. SVM, for instance, excels in binary classification tasks, making it suitable for scenarios where pixelwise classification is required.

5.3.2 Deep learning unveiled

DL, a subset of ML, introduces neural networks with multiple layers (deep neural networks) to automatically learn hierarchical representations of data. CNNs have become pivotal in medical image segmentation [194]. CNNs excel at capturing spatial hierarchies and have proven effective in scenarios with complex anatomical structures. The architecture of CNNs, inspired by the human visual system, involves convolutional layers for feature extraction, pooling layers for down-sampling, and fully connected layers for classification.

5.3.3 Convolutional Neural Networks (CNNs)

CNNs have become synonymous with DL in medical image segmentation. Their architecture enables automatic and adaptive learning of spatial hierarchies from input images. U-Net, a popular CNN architecture, incorporates skip connections to preserve fine details during the down sampling process [195]. This architecture has been particularly successful in medical image segmentation, including tasks such as organ segmentation, tumor detection, and lesion identification.

5.3.4 Recurrent Neural Networks (RNNs) and Long Short-Term Memory (LSTM)

While CNNs excel at spatial understanding, dynamic medical imaging modalities, such as video sequences, require consideration of temporal dependencies. RNNs and Long Short-Term Memory (LSTM) networks, which are specialized types of recurrent networks, are adept at capturing temporal dynamics [196]. In scenarios where temporal information is crucial, such as cardiac imaging or video-based endoscopy, these architectures enhance segmentation accuracy.

5.3.5 Transfer learning and pre-trained models

The application of transfer learning techniques provides a potentially effective approach to tackle the obstacles that arise from the scarcity of annotated data in the field of medical image analysis. Transfer learning is the process of using information acquired from pre-trained models on large datasets and applying it to various tasks or fields with smaller datasets. Transfer learning in medical imaging for noise reduction involves using the obtained predictive power from other datasets to enhance the effectiveness of noise reduction algorithms. Transfer learning allows for the effective transfer of information about low-level image features, noise characteristics, and structural patterns by adjusting pre-trained models with domain-specific medical imaging data. This method speeds up development and improves the capacity of noise reduction algorithms to generalize, especially in situations when obtaining huge annotated medical imaging datasets is not feasible or too expensive. Investigating transfer learning methods has tremendous potential for enhancing noise reduction in medical image processing and enhancing the quality of diagnostic imaging [197, 198].

5.3.6 Generative Adversarial Networks (GANs)

GANs introduce a unique dynamic by involving a generator and discriminator in a competitive learning scenario. In medical image segmentation, GANs contribute by generating realistic synthetic images [199]. This is particularly useful in scenarios with

Methodology	Precision (%)	Recall (%)	Dice Coefficient	Jaccard Index
Traditional Segmentation (Thresholding)	88.2	91.5	0.896	0.834
Traditional Segmentation (Region Growing)	82.6	89.2	0.846	0.773
Traditional Segmentation (Edge Detection)	90.1	88.7	0.892	0.825
Machine Learning (CNN)	95.3	93.8	0.942	0.911
Machine Learning (U-Net)	96.8	94.5	0.958	0.934
Machine Learning (Random Forest)	89.7	90.2	0.904	0.865

 Table 8
 Comparative analysis: Precision, Recall, Dice coefficient, and Jaccard index



Fig. 5 Analysis of Precision for medical image segmentation



Fig. 6 Analysis of Recall for medical image segmentation

limited annotated data for training. GANs have found applications in generating synthetic medical images for augmenting datasets and improving model robustness.

In essence, ML-based segmentation techniques represent a powerful arsenal in the medical imaging domain, offering a spectrum of approaches to cater to diverse challenges and modalities. The evolution of these techniques is fueled by ongoing research, addressing limitations, and adapting to the unique demands of medical image analysis.



Dice Coefficient

Fig. 7 Analysis of Dice Coefficient for medical image segmentation



Jaccard Index



5.4 Comparative analysis

An analysis of both traditional and ML-based methods proves essential in the dynamic sector of medical image segmentation. The purpose of the analysis provided in Table 8 is to provide insights into the strengths, limits, and performance metrics associated with these segmentation methodologies. A comprehensive analysis of key metrics, including Precision, Recall, Dice Coefficient, and Jaccard Index, which were utilized to evaluate the performance of medical image segmentation techniques, is illustrated in Figs. 5, 6, 7, and 8. These metrics play an essential role in evaluating the precision and effectiveness of segmentation techniques, providing valuable insights into different aspects of their operation.

In order to interpret the data visualization, the distances between each of the bars representing each metric for various segmentation methods must be analyzed. Improved levels denote enhanced performance in the corresponding metric, thereby presenting a comprehensive assessment of the advantages and disadvantages of the segmentation methods with respect to precision, recall, Dice Coefficient, and Jaccard Index.

6 Challenges in noise removal and segmentation

6.1 Noise variability

Medical imaging is a diverse field encompassing various modalities, each with unique characteristics that contribute to inherent variability in noise patterns. Effectively addressing this variability presents a substantial challenge in the development of noise removal techniques. Various imaging modalities, including X-ray, MRI, and CT, exhibit distinct noise sources and characteristics [200, 201]. For instance, X-ray images may be afflicted by quantum noise stemming from the statistical nature of X-ray photon interactions, while MRI images may be influenced by thermal noise due to fluctuations in temperature during image acquisition. Crafting noise removal algorithms that can dynamically adapt to the specific noise profile of each modality is essential for achieving optimal results.

The challenge of noise variability extends beyond the inherent characteristics of each imaging modality. Factors such as acquisition settings, patient conditions, and equipment variations further contribute to the complexity of noise patterns [202]. Quantum noise in X-ray imaging may vary based on exposure settings, while thermal noise in MRI could be influenced by the magnetic field strength.

To address noise variability comprehensively, researchers are exploring adaptive algorithms that can analyze and learn the specific noise characteristics inherent in different imaging scenarios. ML techniques, including DL models, are being employed to create noise removal algorithms capable of adapting to the nuances of each modality, providing a more tailored and effective approach to noise reduction [203]. Navigating the intricate landscape of noise variability is paramount for advancing the accuracy and reliability of medical image analysis across diverse imaging modalities. The Table 9 outlines the aspects of the challenges related to noise variability of imaging modalities,

Imaging Modality	Quantum Noise	Electronic Noise	Radiation Interfer- ence	Tempera- ture Varia- tions
X-ray [204]	✓	~	×	×
MRI [205]	×	×	×	✓
CT [206]	✓	×	\checkmark	×
Ultrasound [207]	×	✓	×	×
Nuclear Medicine [208]	×	\checkmark	×	×
PET [209]	×	✓	×	×
Mammography [210]	\checkmark	\checkmark	×	×
Fluoroscopy [211]	\checkmark	✓	×	×

 Table 9 Noise variability in medical imaging

each with parameters indicating the presence or absence of Quantum Noise, Electronic Noise, Radiation Interference, and Temperature Variations.

6.2 Anatomical variations

The intricate variability in anatomical structures across patients introduces a significant level of complexity in the domain of medical image segmentation. Anatomical structures can exhibit diverse shapes, sizes, and textures, influenced by factors such as patient age, gender, and health conditions [212]. Successfully addressing anatomical variations is paramount for accurate and reliable image segmentation, a crucial step in medical image analysis.

Segmentation algorithms face the challenge of accommodating the inherent diversity in anatomical structures to precisely delineate regions of interest. For instance, when dealing with abdominal CT scans, the task of segmenting organs becomes particularly challenging due to the inherent variations in organ shapes and positions among different individuals [213]. The liver, for example, may exhibit considerable differences in size, shape, and location from one patient to another.

To overcome the obstacles posed by anatomical variations, researchers are exploring advanced segmentation approaches, frequently incorporating ML and DL techniques [214]. These methodologies aim to develop robust segmentation models that can adapt to the intricacies of anatomical diversity. ML models, particularly CNNs, have shown promise in learning complex patterns and variations, making them suitable for accurate segmentation tasks.

Despite advancements, tackling anatomical variations remains a persistent obstacle in the field. Collaborative efforts between medical professionals, image processing experts, and ML researchers are crucial for developing segmentation models that can accommodate the inherent complexity of anatomical structures across diverse patient populations [215]. The ultimate goal is to enhance the precision and reliability of medical image segmentation for improved diagnosis and treatment planning. Table 10 outlines the challenges associated with anatomy variations using various imaging techniques.

Challenge	X-ray	MRI	СТ	Ultrasound	Nuclear Medi- cine	PET	Mammography	Fluoroscopy
Diverse organ shapes [216]	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Variations in organ sizes [217]	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Differences in organ textures [218]	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Patient age influence [219]	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Gender-related variations [220]	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table 10 Anatomical Variations across different imaging modalities

6.3 Lack of ground truth data

The reliance on annotated ground truth data for training ML models is fundamental to the development of accurate and robust noise removal and segmentation techniques. However, the limited availability of comprehensive and accurately annotated datasets regularly hinders this reliance. Acquiring ground truth data for medical images involves meticulous annotation by domain experts, a process that is not only time-consuming but also resource-intensive [221].

The annotation process requires skilled professionals who can accurately label images to serve as reference points for training ML models. Discrepancies in annotations may arise due to variations in interpretation among different experts, introducing ambiguity and potential challenges in model training. In addition, the lack of diverse and well-annotated datasets is a big problem that makes it hard to train ML models that work well in real-life situations [222].

Addressing the challenge of a lack of ground truth data requires collaborative efforts across disciplines. Medical professionals, image processing experts, and ML practitioners need to work together to create standardized and comprehensive datasets [223]. Initiatives such as data-sharing collaborations and the development of annotation guidelines can contribute to building datasets that accurately represent the complexities of medical images.

Advancements in addressing the lack of ground truth data are pivotal for unlocking the full potential of noise removal and segmentation methodologies in the realm of medical image analysis and diagnosis. Future progress relies on the commitment to overcoming these challenges through interdisciplinary collaboration and the development of robust, well-annotated datasets that reflect the diverse and nuanced nature of medical images [224]. Dataset availability, annotation process complexity, resource intensity, discrepancies in expert annotations, as well as potential ambiguity in the annotated data, are some of the factors that are emphasized in Table 11, which gives an organized overview of the issues that are associated with the absence of ground truth data.

7 Future research directions

Researchers are investigating the potential of advanced ML models, including DL architectures such as CNNs and RNNs, to enhance medical imaging. The present study aims to use advanced algorithms to improve noise reduction and segmentation in medical imaging, which leads to more accurate and comprehensive diagnostic assessments. A novel field of study is the combination of data from numerous imaging techniques, including X-ray, MRI, and CT, using multimodal data fusion. The aim of this effort is to improve the accuracy and comprehensiveness of evaluations by using the unique benefits of each modality. There is more emphasis on the need for interpretable AI models in healthcare. Researchers are investigating ways to improve the transparency and interpretability of AI models in medical imaging so that medical professionals can comprehend and depend on the assessments made by these sophisticated systems. Also, emphasizing the essential importance of immediate data in clinical decision-making, there is a focused endeavor to investigate methods and technology for attaining real-time segmentation in medical imaging. This improvement can provide rapid and precise information, greatly influencing clinical decision support procedures. These research issues aim to expand medical imaging by using advanced

ChallengeMitigation StrategiesCollaboration InitiativesImpact on Model Generali- zationData Sharing Guid Model Generali- zationLimited availability of comprehen- sive datasets [225]Encourage multi-institutional col- laborationsEstablish data-sharing platformsHighData Sharing GuidLimited availability of comprehen- sive datasets [225]Encourage multi-institutional col- laborationsEstablish data-sharing platformsHighStandardize annotaLime-consuming annotation processInvest in automated annotation toolsFoster interdisciplinary collabora- tionsModerateDevelop annotatior(226]Invest in automated annotation for annotationSupport joint efforts between aca- demia and industryHighDefine resource-eff norms(228]assessmentsPromote on industryModerate demia and industryModerate toomsDefine resource-eff source-intersive annotation(228]assessmentsPromote on going communicationModerate demia and industryModerate mortsEstablish consensu(228]annotationDevelop clear annotation guidelinesPromote on going communicationModerate mortsEstablish consensu(228]annotation guidelinesPromote on going communicationModerateEstablish consensu(228]annotation guidelinesPromote on going communicationModerateEstablish consensu(228]annotation guidelinesPromote on going communicationModerateEncorage continu	Table 11 Lack of ground truth data ch	allenges			
Limited availability of comprehen- sive datasets [225]Encourage multi-institutional col- laborationsEstablish data-sharing platformsHighStandardize annotaTime-consuming annotation processInvest in automated annotation toolsFoster interdisciplinary collabora- tionsModerateDevelop annotation[226]Resource-intensive annotationOptimize resource allocation for annotationSupport joint efforts between aca- demia and industryHighDefine resource-eff[226]Resource-intensive annotationOptimize resource allocation for annotationSupport joint efforts between aca- demia and industryHighDefine resource-eff[228]Bescource-intensive annotationsImplement inter-rater reliabilityFacilitate cross-disciplinary trainingModerate moresBeronseu works[228]Develop clear annotation guidelinesPromote ongoing communicationModerateEncourage continu	Challenge	Mitigation Strategies	Collaboration Initiatives	Impact on Model Generali- zation	Data Sharing Guidelines
Time-consuming annotation processInvest in automated annotation toolsFoster interdisciplinary collabora-ModerateDevelop annotation[226][226]innostinnostinnostinnostinnost[226]Resource-intensive annotationOptimize resource allocation forSupport joint efforts between aca-HighDefine resource-effResource-intensive annotationOptimize resource allocation forSupport joint efforts between aca-HighDefine resource-effprocess [227]annotationOptimize resource allocation forSupport joint efforts between aca-HighDefine resource-effprocess [227]annotationInplement inter-rater reliabilityFacilitate cross-disciplinary trainingModerateEstablish consensu[228]assessmentsprogramsprogramsModerateBrownisworksAmbiguity in annotated data [229]Develop clear annotation guidelinesPromote ongoing communicationModerateEncourage continumentsannotation in the inter-rate reliabilityPromote ongoing communicationModerateEncourage continuments	Limited availability of comprehen- sive datasets [225]	Encourage multi-institutional col- laborations	Establish data-sharing platforms	High	Standardize annotation protocols
Resource-intensive annotationOptimize resource allocation for annotationSupport joint efforts between aca- demia and industryHighDefine resource-eff normsprocess [227]annotationannotationdemia and industryDefine resource-effnormsDiscrepancies in expert annotationsImplement inter-rater reliabilityFacilitate cross-disciplinary trainingModerateEstablish consensu[228]assessmentsprogramsworksworksAmbiguity in annotated data [229]Develop clear annotation guidelinesPromote ongoing communicationModerateEncourage continu- ments	Time-consuming annotation process [226]	Invest in automated annotation tools	Foster interdisciplinary collabora- tions	Moderate	Develop annotation efficiency guide- lines
Discrepancies in expert annotationsImplement inter-rater reliabilityFacilitate cross-disciplinary trainingModerateEstablish consensu-[228]assessmentsprogramsworksworksAmbiguity in annotated data [229]Develop clear annotation guidelinesPromote ongoing communicationModerateEncourage continu-annotated data [229]Develop clear annotation guidelinesPromote ongoing communicationModerateEncourage continu-annotated data [229]Develop clear annotation guidelinesPromote ongoing communicationModerateEncourage continu-	Resource-intensive annotation process [227]	Optimize resource allocation for annotation	Support joint efforts between aca- demia and industry	High	Define resource-efficient annotation norms
Ambiguity in annotated data [229] Develop clear annotation guidelines Promote ongoing communication Moderate Encourage continue among annotators more more ments	Discrepancies in expert annotations [228]	Implement inter-rater reliability assessments	Facilitate cross-disciplinary training programs	Moderate	Establish consensus-building frame- works
	Ambiguity in annotated data [229]	Develop clear annotation guidelines	Promote ongoing communication among annotators	Moderate	Encourage continuous quality assessments

challenges
data
truth
ground
of
Lack
Ξ

Research Issue	Description
Ensuring the scalability of AI models in diverse settings	Evaluating the scalability of AI models to guarantee their efficiency and performance in various health- care environments and organizations
Investigating the potential biases in AI predictions	Identifying and addressing any biases in AI predic- tions to guarantee fair and equal results for every group of patients
Studying the impact of AI on healthcare disparities	Evaluating the impact of using AI models in medical imaging on healthcare inequalities and aiming for equitable solutions
Addressing data privacy concerns in AI applica- tions	Developing strong strategies to tackle data privacy issues related to the use of sensitive medical data in AI applications
Studying the ethical implications of AI-driven deci- sion support	Exploring the ethical consequences of depending on AI-powered decision help in medical imaging and establishing ethical use protocols
Investigating the influence of AI on radiologist training	Investigating the potential effects of AI technology on the education and skill enhancement of radiologists and other healthcare practitioners
Developing frameworks for transparent AI decision- making	Providing transparent frameworks to help healthcare practitioners comprehend the decision-making process of AI models in medical imaging
Investigating the impact of AI on patient outcomes	Evaluating the effects of using AI in medical imag- ing on patient results, treatment effectiveness, and healthcare quality
Lack of annotated datasets for advanced models	Insufficient availability of annotated datasets suitable for training and verifying sophisticated ML models
Generalization challenges in diverse medical images	Issues for developing models that can effectively generalize over a wide range of medical images with various features
Limited interpretability of DL models	Issues in understanding the decisions made by DL models are impeding their use in medical applica- tions
Integration challenges in multimodal data fusion	Issues in effectively combining data from several imaging techniques to obtain valuable information
Ethical considerations in AI applications	Exploring ethical, privacy, and bias issues connected with utilizing AI models in medical imaging
Ensuring model robustness in noisy environments	Improving the resilience of AI models in medical imaging in noisy environments and unpredictable circumstances
Cybersecurity concerns in AI applications	Maintaining patient data and privacy by addressing cybersecurity issues related to using AI models in healthcare
Investigating the impact of AI on healthcare workflows	Studying the impact of using AI models on present healthcare processes and identifying opportunities for improvement
Assessing long-term reliability of AI models	Evaluating the enduring dependability and steadfast- ness of AI models to guarantee sustained perfor- mance over prolonged durations
Exploring edge computing for real-time processing	Addressing the practicality of implementing AI models on edge devices for immediate processing in contexts with limited resources

Table 12 Research issues and challenges in the field of medical imaging

Table 12 (continued)

Research Issue	Description
Evaluating the cost-effectiveness of AI adoption	Investigating the economic consequences and effi- ciency of using AI technology in medical imaging processes
Addressing regulatory compliance and patient consent	Ensuring adherence to regulations and gaining informed permission from patients for the use of AI in medical imaging
Establishing robust model validation procedures	Providing standardized and rigorous protocols for validating and assessing ML models in medical imaging
Bridging the gap between AI research and clinical practice	Facilitating in the implementation of AI research outcomes in healthcare environments
Investigating the impact of AI on diagnostic accuracy	Evaluating the impact of incorporating AI models on the diagnostic precision and decision-making skills of healthcare practitioners
Developing mechanisms for continuous model improvement	Providing methods to enhance models continuously by incorporating input, adapting to evolving medi- cal practices, and integrating new data
Studying the psychological impact on healthcare professionals	Exploring the psychological effects of AI implemen- tation on healthcare workers and resolving issues around employment responsibilities and obligations
Investigating the impact of AI on patient-doctor communication	Investigating the impact of AI integration in medical imaging on the interaction between healthcare providers and patients
Enhancing model interpretability for end-users	Developing methods to improve the interpretability of AI models, allowing medical professionals to have confidence in and comprehend the assessments generated by the model
Exploring unsupervised learning for limited anno- tated data	Exploring the use of unsupervised learning methods to make the most of scarce annotated data in medi- cal imaging
Real-time processing constraints	Implementing algorithms that enable segmentation in real-time while considering advantage of comput- ing limitations in clinical applications
Standardization of annotation protocols	Absence of specific requirements for annotating medical images causes variability in annotations and expected discrepancies
Lack of collaboration between academia and industry	Limited collaboration between academic researchers and industry specialists is impeding the transforma- tion of research into practical solutions
Adapting models to evolving medical imaging tech	Ensuring that ML models are able to adapt to emerg- ing technologies and improvements in medical imaging
Incorporating user feedback into model develop- ment	Providing solutions that integrate input from health- care experts in order to enhance the usability and efficacy of AI models
Addressing bias and fairness in AI models	Addressing and reducing biases in AI models to guarantee equitable and impartial results, especially in varied patient demographics
Regulatory challenges in AI adoption	Addressing regulatory challenges while establishing criteria for using AI models in medical imaging

Table 12 (continued)	
Research Issue	Description
Development of user-friendly AI interfaces	Provide interfaces that are intuitive and readily understandable for healthcare providers with less technological knowledge

ML, combining several modes of data, enhancing interpretability, and enabling real-time capabilities to enhance healthcare outcomes. The research challenges and issues in medical imaging are presented in Table 12. These concerns and problems encompass a variety of factors, such as advancements in technology, ethical implications, adherence to regulations, and the effective incorporation of artificial intelligence into healthcare procedures.

8 Conclusion

Multimedia Tools and Applications

In summary, this paper has explored the intricate landscape of noise removal and segmentation techniques in medical imaging, delving into both traditional and advanced ML-based approaches. The investigation encompassed the significance of medical imaging in healthcare, highlighting its pivotal role in diagnosis and treatment. The research objectives centered on comparing the efficacy of traditional and ML-based methods for noise removal and segmentation. The paper meticulously navigated through various medical imaging modalities, elucidating their principles and applications. It provided insights into the challenges posed by noise variability, anatomical variations, and the scarcity of ground truth data in the medical imaging domain. The findings underscored the evolution from conventional noise reduction techniques to the promising realm of ML-based approaches. Comparative analyses shed light on the strengths and limitations of each methodology. The role of segmentation in medical image analysis was thoroughly discussed, emphasizing its importance in enhancing diagnostic precision. Challenges related to noise variability, anatomical variations, and the lack of ground truth data were dissected, recognizing the complexity of these hurdles. Looking forward, the paper proposed future research directions, advocating for the exploration of advanced ML models, multimodal data fusion, interpretable AI, and realtime segmentation. These directions aim to push the boundaries of medical imaging, promising more accurate, efficient, and real-time diagnostic capabilities. The implications for medical imaging are profound. The integration of advanced ML models holds the potential to revolutionize noise removal and segmentation techniques, paving the way for more accurate and timely diagnoses. The exploration of multimodal data fusion addresses the need for comprehensive diagnostic analyses, considering the unique strengths of different imaging modalities. The call for interpretable AI models responds to the crucial demand for transparency and trust in AI-driven decision-making, ensuring seamless integration into clinical workflows. Real-time segmentation emerges as a key component for prompt and effective clinical decision support. In closing, the significance of effective noise removal and segmentation in medical imaging cannot be overstated. These techniques are not mere technical processes but integral components that directly impact diagnostic accuracy, treatment planning, and patient outcomes. The continual evolution and integration of advanced methodologies, as outlined in this paper, underscore the dynamic nature of medical imaging research and its pivotal role in shaping the future of healthcare.

Author contributions In this study, R.R.K. and R.P. all contributed significantly to the research efforts. R.R.K. played a key role in the execution of the experiments, while R.P. contributed to various aspects of the research process. The collaborative efforts of R.R.K. and R.P. are evident in the combined writing and development of the paper.

Funding Not applicable.

Data availability Data sharing not applicable to this article as no datasets were generated or analyzed during the current study.

Declarations

Ethics approval and consent to participate Not applicable.

Consent for publication Not applicable.

Conflict of interest The authors have no conflict of interest to declare that are relevant to the content of this article.

References

- 1. Li D (2014) A tutorial survey of architectures, algorithms, and applications for deep learning. APSIPA Trans Signal Inf Process 3
- 2. Hinton GE (2006) Reducing the dimensionality of data with neural networks. Sci. 313:504-507
- 3. Bengio Y, Lamblin P, Popovici D, Larochelle H (2007) Greedy layer-wise training of deep networks. Adv Neural Inf Process Syst 19
- 4. Silver D (2016) Mastering the game of Go with deep neural networks and tree search. Nature 529:484-489
- 5. Hinton RSZ, Geoffrey E (1994) Autoencoders, minimum description length and Helmholtz free energy. Adv Neural Inf Process Syst 3
- 6. Hinton GE, Salakhutdinov RR (2006) Reducing the dimensionality of data with neural networks. Sci. 313:5786
- Yap MH et al (2017) Automated breast ultrasound lesions detection using convolutional neural networks. IEEE J Biomed Heal Inf 22:1218–1226
- Jiménez-Sánchez A, Tardy M, Ballester MAG, Mateus D, Piella G (2023) Memory-aware curriculum federated learning for breast cancer classification. Comput Methods Prog Biomed 229:107318
- 9. Karthik R, Menaka R, Siddharth M (2022) Classification of breast cancer from histopathology images using an ensemble of deep multiscale networks. Biocybern Biomed Eng 42:963–976
- Ravelli A et al (2015) Breast cancer circulating biomarkers: advantages, drawbacks, and new insights. Tumor Biol 36:6653–65
- Houssein EH, Emam MM, Ali AA (2022) An optimized deep learning architecture for breast cancer diagnosis based on improved marine predators algorithm. Neural Comput Appl 34(18015):18033
- Priyadarshi R, Gupta B (2023) 2-D coverage optimization in obstacle-based FOI in WSN using modified PSO. J Supercomput 79(5):4847–4869. https://doi.org/10.1007/s11227-022-04832-6
- Priyadarshi R, Rawat P, Nath V (2019) Energy dependent cluster formation in heterogeneous wireless sensor network. Microsyst Technol 25(6):2313–2321. https://doi.org/10.1007/s00542-018-4116-7
- 14 Piantadosi G, Sansone M, Fusco R, Sansone C (2020) Multi-planar 3d breast segmentation in MRI via deep convolutional neural networks. Artif Intell Med 103:101781
- 15 Oloomi M, Moazzezy N, Bouzari S (2020) Comparing blood versus tissue-based biomarkers expression in breast cancer patients. Heliyon 6:e03728
- Rawat P, Chauhan S, Priyadarshi R (2020) Energy-efficient clusterhead selection scheme in heterogeneous wireless sensor network. J Circ Syst Comput 29(13):2050204. https://doi.org/10.1142/S0218 126620502047
- 17. Nissan N, Bauer E, Moss Massasa EE, Sklair-Levy M (2022) Breast MRI during pregnancy and lactation: clinical challenges and technical advances. Insights Imaging 13:71
- Lu W, Li Z, Chu J (2017) A novel computer-aided diagnosis system for breast MRI based on feature selection and ensemble learning. Comput Biol Med 83:157–165

- Salama WM, Elbagoury AM, Aly MH (2020) Novel breast cancer classification framework based on deep learning. IET Image Proc 14,
- Ramadan SZ (2020) Methods used in computer-aided diagnosis for breast cancer detection using mammograms: a review. J Heal Eng 2020:9162464
- Pandey A, Kumar D, Priyadarshi R, Nath V (2023) Development of smart village for better lifestyle of farmers by crop and health monitoring system. In V. Nath & J. K. Mandal (Eds.), Lecture Notes in Electrical Engineering (Vol. 887, pp. 689–694). Springer Nature Singapore. https://doi.org/10.1007/ 978-981-19-1906-0_57
- 22. Mohanta BK, Jena D, Mohapatra N, Ramasubbareddy S, Rawal BS (2022) Machine learning based accident prediction in secure IoT enable transportation system. J Intell Fuzzy Syst 42(713):725
- 23 Matsumoto Y, Katsumura A, Miki N (2022) Pressure-controlled ultrasound probe for reliable imaging in breast cancer diagnosis. Jpn J Appl Phys 61:SD1035
- Priyadarshi R, Gupta B, Anurag A (2020) Deployment techniques in wireless sensor networks: a survey, classification, challenges, and future research issues. J Supercomput 76(9):7333–7373. https://doi.org/10.1007/s11227-020-03166-5
- 25 Sethy PK, Behera SK (2022) Automatic classification with concatenation of deep and handcrafted features of histological images for breast carcinoma diagnosis. Multimed Tools Appl 81:9631–9643
- Priyadarshi R, Soni SK, Nath V (2018) Energy efficient cluster head formation in wireless sensor network. Microsyst Technol 24(12):4775–4784. https://doi.org/10.1007/s00542-018-3873-7
- Anurag A, Priyadarshi R, Goel A, Gupta B (2020) 2-D coverage optimization in WSN using a novel variant of particle swarm optimisation. 2020 7th International Conference on Signal Processing and Integrated Networks, SPIN 2020, 663–668. https://doi.org/10.1109/SPIN48934.2020.9070978
- 28 Al Ewaidat H, Ayasrah M (2022) A concise review on the utilization of abbreviated protocol breast MRI over full diagnostic protocol in breast cancer detection. Int J Biomed Imaging 2022:1–8
- 29 Sahiner B et al (2007) Malignant and benign breast masses on 3d us volumetric images: effect of computer-aided diagnosis on radiologist accuracy. Radiology 242:716–724
- 30 Petrova D et al (2022) Duration of the patient interval in breast cancer and factors associated with longer delays in low-and middle-income countries: a systematic review with meta-analysis. Psychooncology. 32:13–24
- Priyadarshi R, Gupta B (2021) Area Coverage Optimization in Three-Dimensional Wireless Sensor Network. Wireless Pers Commun 117(2):843–865. https://doi.org/10.1007/s11277-020-07899-7
- Taheri S, Golrizkhatami Z (2022) Magnification-specific and magnification-independent classification of breast cancer histopathological image using deep learning approaches. Signal Image Video Process 2022
- 33. Joseph C et al (2018) Breast cancer intratumour heterogeneity: current status and clinical implications. Histopathology 73:717–731
- 34. Gonçalves CB, Souza JR, Fernandes H (2022) CNN architecture optimization using bio-inspired algorithms for breast cancer detection in infrared images. Comput Biol Med 142:105205
- 35. Krithiga R, Geetha P (2021) Breast cancer detection, segmentation and classification on histopathology images analysis: a systematic review. Arch Comput Methods Eng 28:2607–2619
- 36 Ruan D, Sun L (2022) Diagnostic performance of PET/MRI in breast cancer: a systematic review and Bayesian bivariate meta-analysis. Clin Breast Cancer 23(108):124
- 37 Jeleń Ł, Krzyżak A, Fevens T, Jeleń M (2016) Influence of feature set reduction on breast cancer malignancy classification of fine needle aspiration biopsies. Comput Biol Med 79:80–91
- 38 ElOuassif B, Idri A, Hosni M, Abran A (2021) Classification techniques in breast cancer diagnosis: a systematic literature review. Comput Methods Biomech Biomed Eng 9:50–77
- Priyadarshi R, Nath V (2019) A novel diamond–hexagon search algorithm for motion estimation. Microsyst Technol 25(12):4587–4591. https://doi.org/10.1007/s00542-019-04376-5
- 40 Lee J, Kang BJ, Park GE, Kim SH (2022) The usefulness of magnetic resonance imaging (MRI) for the detection of local recurrence after mastectomy with reconstructive surgery in breast cancer patients. Diagnostics 12:2203
- 41 Mehrotra R, Yadav K (2022) Breast cancer in India: present scenario and the challenges ahead. World J Clin Oncol 13:209–218
- 42 Wu YC et al (1995) Classification of microcalcifications in radiographs of pathologic specimens for the diagnosis of breast cancer. Acad Radiol 2:199–104
- 43. Mohamed A et al (2022) The impact of data processing and ensemble on breast cancer detection using deep learning. J Comput Commun 1
- Priyadarshi R, Rana H, Srivastava A, Nath V (2023) A Novel Approach for Sink Route in Wireless Sensor Network. In V. Nath & J. K. Mandal (Eds.), Lecture Notes in Electrical Engineering (Vol. 887, pp. 695–703). Springer Nature Singapore. https://doi.org/10.1007/978-981-19-1906-0_58

- Priyadarshi R, Singh L, Randheer, Singh A (2018) A Novel HEED Protocol for Wireless Sensor Networks. 2018 5th International Conference on Signal Processing and Integrated Networks, SPIN 2018, 296–300. https://doi.org/10.1109/SPIN.2018.8474286
- 46. Singh VK et al. (2020) Breast tumor segmentation and shape classification in mammograms using generative adversarial and convolutional neural network. Expert SystAppl 139
- 47. Trister AD, Buist DS, Lee CI (2017) Will machine learning tip the balance in breast cancer screening? JAMA Oncol 3
- 48. Hu Q et al (2021) Improved classification of benign and malignant breast lesions using deep feature maximum intensity projection MRI in breast cancer diagnosis using dynamic contrast-enhanced MRI. Radiology 3
- 49. Galati F et al (2022) Radiologic-pathologic correlation in breast cancer: do MRI biomarkers correlate with pathologic features and molecular subtypes? Eur Radiol Exp 6:39
- 50. Mann RM et al (2022) Breast cancer screening in women with extremely dense breasts recommendations of the European society of breast imaging (eusobi). Eur Radiol 32:4036–4045
- 51. Zeiser FA et al (2021) Deepbatch: a hybrid deep learning model for interpretable diagnosis of breast cancer in whole-slide images. Expert Syst Appl 185:115586
- 52. Pramanik R, Pramanik P, Sarkar R (2023) Breast cancer detection in thermograms using a hybrid of GA and GWO based deep feature selection method. Expert Syst Appl 219
- 53. Spanhol FA, Oliveira LS, Petitjean C, Heutte L (2015) A dataset for breast cancer histopathological image classification. IEEE Trans Biomed Eng 63
- Pal UM et al (2021) Hybrid spectral-irdx: near-ir and ultrasound attenuation system for differentiating breast cancer from adjacent normal tissue. IEEE Trans Biomed Eng 68:3554–3563
- 55. Joseph AA, Abdullahi M, Junaidu SB, Ibrahim HH, Chiroma H (2022) Improved multi-classification of breast cancer histopathological images using handcrafted features and deep neural network (dense layer). Intell Syst Appl 14:200066
- 56. Barrios CH (2022) Global challenges in breast cancer detection and treatment. Breast 62
- Shen L et al (2019) Deep learning to improve breast cancer detection on screening mammography. Sci Rep 9:3–6
- Ahmadian S, Ahmadian M, Jalili M (2022) A deep learning based trust-and tag-aware recommender system. Neurocomputing 488:557–571
- 59. Moreira IC et al (2012) Inbreast: toward a full-field digital mammographic database. Acad Radiol 19:236–48
- Thompson JL, Wright GP (2021) The role of breast MRI in newly diagnosed breast cancer: an evidence-based review. Am J Surg 221:525–528
- 61. Momenimovahed Z, Salehiniya H (2019) Epidemiological characteristics of and risk factors for breast cancer in the world. Breast Cancer Targets Ther 11:151–164
- 62. Yusuf A, Okafor I, Olubodun T, Onigbogi O (2022) Breast cancer knowledge and screening practices among undergraduates in a Nigerian tertiary institution, southwest region. Afr Heal Sci 4
- 63. Hussein H et al. (2023) Supplemental breast cancer screening in women with dense breasts and negative mammography: a systematic review and meta-analysis. Radiology 306
- 64. Koh J, Yoon Y, Kim S, Han K, Kim E-K (2022) Deep learning for the detection of breast cancers on chest computed tomography. Clin Breast Cancer 22:26–31
- Saber A, Sakr M, Abo-Seida OM, Keshk A, Chen H (2021) A novel deep-learning model for automatic detection and classification of breast cancer using the transfer-learning technique. IEEE Access 9:71194–71209
- 66. Boersma L et al (2022) Model-based selection for proton therapy in breast cancer: development of the national indication protocol for proton therapy and first clinical experiences. Clin Oncol 34:88–94
- 67. Wang X et al (2022) Intelligent hybrid deep learning model for breast cancer detection. Electronics 11:2767
- Volterrani L et al (2020) Dual-energy CT for locoregional staging of breast cancer: preliminary results. Am J Roentgenol 214
- Priyadarshi R, Singh L, Singh A, Thakur A (2018) SEEN: Stable Energy Efficient Network for Wireless Sensor Network. 2018 5th International Conference on Signal Processing and Integrated Networks, SPIN 2018, 338–342. https://doi.org/10.1109/SPIN.2018.8474228
- Suckling J (1994) The mammographic images analysis society digital mammogram database. Exerpta Medica 1069:236–248
- 71. Yang X et al (2020) Deep learning signature based on staging CT for preoperative prediction of sentinel lymph node metastasis in breast cancer. Acad Radiol 27:1226–1233
- 72. Castro-Tapia S et al (2021) Classification of breast cancer in mammograms with deep learning adding a fifth class. Appl Sci 11:11398

- 73. Hadebe B, Harry L, Ebrahim T, Pillay V, Vorster M (2023) The role of PET/CT in breast cancer. Diagnostics 13:429–437
- 74. Sahu A, Das PK, Meher S (2023) High accuracy hybrid CNN classifiers for breast cancer detection using mammogram and ultrasound datasets. Biomed Signal Process Control 80:104292
- Yassin NI, Omran S, Houby EM, Allam H (2018) Machine learning techniques for breast cancer computer aided diagnosis using different image modalities: a systematic review. Comput Methods Prog Biomed 156:25–45
- 76 Aslan MF (2023) A hybrid end-to-end learning approach for breast cancer diagnosis: convolutional recurrent network. Comput Electr Eng 105:108562
- Priyadarshi R, Rawat P, Nath V, Acharya B, Shylashree N (2020) Three level heterogeneous clustering protocol for wireless sensor network. Microsyst Technol 26(12):3855–3864. https://doi.org/10. 1007/s00542-020-04874-x
- 78 Bouron C et al (2022) Prognostic value of metabolic, volumetric and textural parameters of baseline [18f] FDG PET/CT in early triple-negative breast cancer. Cancers (Basel). 14:637
- 79 Dhillon A, Singh A (2020) ebrecap: extreme learning-based model for breast cancer survival prediction. IET Syst Biol 14:160–169
- Desai S, Kanphade R, Priyadarshi R, Rayudu KVBV, Nath V (2023) A novel technique for detecting crop diseases with efficient feature extraction. IETE J Res, 1–9.https://doi.org/10.1080/03772063. 2023.2220667
- 81. Chen X et al. (2020) CNN-based quality assurance for automatic segmentation of breast cancer in radiotherapy. Front Oncol 10
- Priyadarshi R, Bhardwaj P, Gupta P, Nath V (2023) Utilization of smartphone-based wireless sensors in agricultural science: A State of Art. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical* Engineering (Vol. 887, pp. 681–688). Springer Nature Singapore. https://doi.org/10.1007/978-981-19-1906-0_56
- 83 Nassif AB, Talib MA, Nasir Q, Afadar Y, Elgendy O (2022) Breast cancer detection using artificial intelligence techniques: a systematic literature review. Artif Intell Med 127:102276
- 84 Araújo T et al (2017) Classification of breast cancer histology images using convolutional neural networks. PLoS One 12:e0177544
- Priyadarshi R, Singh A, Agarwal D, Verma UC, Singh A (2023) Emerging Smart Manufactory: Industry 4.0 and Manufacturing in India: The Next Wave. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical Engineering* (Vol. 887, pp. 353–363). Springer Nature Singapore. https://doi.org/ 10.1007/978-981-19-1906-0_32
- Moon WK et al (2020) Computer-aided diagnosis of breast ultrasound images using ensemble learning from convolutional neural networks. Comput Methods Prog Biomed 190:106271
- 87. Pan P et al (2021) Tumor segmentation in automated whole breast ultrasound using bidirectional ISTM neural network and attention mechanism. Ultrasonics 110:106271
- Gupta T, Kumar A, Priyadarshi R (2020) A novel hybrid precoding technique for millimeter wave. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical Engineering* (Vol. 642, pp. 481–493). Springer Singapore. https://doi.org/10.1007/978-981-15-2854-5_42
- Singh C, Imam T, Wibowo S, Grandhi S (2022) A deep learning approach for sentiment analysis of covid-19 reviews. Appl Sci 12:3709
- Satesh VA, Kumar A, Priyadarshi R, Nath V (2021) A novel deployment scheme to enhance the coverage in wireless sensor network. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes* in *Electrical Engineering* (Vol. 673, pp. 985–993). Springer Singapore. https://doi.org/10.1007/978-981-15-5546-6_82
- Hamed G, Marey M, Amin SE, Tolba MF (2021) Automated breast cancer detection and classification in full field digital mammograms using two full and cropped detection paths approach. IEEE Access 9:116898–116913
- 92. Tong L, Mitchel J, Chatlin K, Wang MD (2020) Deep learning based feature-level integration of multi-omics data for breast cancer patients survival analysis. BMC Med Inf. Decis Mak 20:225
- 93. Wang Q et al (2022) Performance of novel deep learning network with the incorporation of the automatic segmentation network for diagnosis of breast cancer in automated breast ultrasound. Eur Radiol 32:7163–7172
- 94. Trang NTH, Long KQ, An PL, Dang TN (2023) Development of an artificial intelligence-based breast cancer detection model by combining mammograms and medical health records. Diagnostics 13:346
- Singh L, Kumar A, Priyadarshi R (2020) Performance and comparison analysis of image processing based forest fire detection. In V. Nath & J. K. Mandal (Eds.), Lecture Notes in Electrical Engineering (Vol. 642, pp. 473–479). Springer Singapore. https://doi.org/10.1007/978-981-15-2854-5_41

- Mehra R (2018) Breast cancer histology images classification: training from scratch or transfer learning? ICT Express 4:247–254
- 97. Huang Q, Chen Y, Liu L, Tao D, Li X (2019) On combining biclustering mining and adaboost for breast tumor classification. IEEE Trans Knowl Data Eng 32:728–738
- Dewangan KK, Dewangan DK, Sahu SP, Janghel R (2022) Breast cancer diagnosis in an early stage using novel deep learning with hybrid optimization technique. Multimed Tools Appl 81:13935–13960
- 99. Graham LJ et al (2014) Current approaches and challenges in monitoring treatment responses in breast cancer. J Cancer 5
- Shim S et al (2023) Radiation dose estimates based on Monte Carlo simulation for spiral breast computed tomography imaging in a large cohort of patients. Med Phys 50:2417–2428
- 101. Priyadarshi R, Yadav S, Bilyan D (2019) Performance analysis of adapted selection based protocol over LEACH protocol. In A. K. Luhach, K. B. G. Hawari, I. C. Mihai, P.-A. Hsiung, & R. B. Mishra (Eds.), *Smart Computational Strategies: Theoretical and Practical Aspects* (pp. 247–256). Springer Singapore. https://doi.org/10.1007/978-981-13-6295-8_21
- 102. Bruckmann NM et al (2021) Prospective comparison of the diagnostic accuracy of 18f-fdg PET/ MRI, MRI, CT, and bone scintigraphy for the detection of bone metastases in the initial staging of primary breast cancer patients. Eur Radiol 31:8714–8724
- Kumar S, Soni SK, Randheer, Priyadarshi R (2020). Performance analysis of novel energy aware routing in wireless sensor network. In: V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical Engineering* (Vol. 642, pp. 503–511). Springer Singapore. https://doi.org/10.1007/978-981-15-2854-5_44
- 104 Thawani R et al (2022) Quantitative DCE-MRI prediction of breast cancer recurrence following neoadjuvant chemotherapy: a preliminary study. BMC Med Imaging 22:182
- 105 Shim S et al (2022) Fully automated breast segmentation on spiral breast computed tomography images. J Appl Clin Med Phys 23:e13726
- 106 Pérez-Benito FJ et al (2020) A deep learning system to obtain the optimal parameters for a threshold-based breast and dense tissue segmentation. Comput Methods Prog Biomed 195:105668
- 107. Priyadarshi, R., Yadav, S., & Bilyan, D. (2019). Performance and comparison analysis of MIEEP routing protocol over adapted LEACH protocol. In A. K. Luhach, K. B. G. Hawari, I. C. Mihai, P.-A. Hsiung, & R. B. Mishra (Eds.), *Smart Computational Strategies: Theoretical and Practical Aspects* (pp. 237–245). Springer Singapore. https://doi.org/10.1007/978-981-13-6295-8_20
- Zipkin RJ et al (2022) Rural-urban differences in breast cancer surgical delays in medicare beneficiaries. Ann Surg Oncol 29:5759–5769
- Debelee TG, Schwenker F, Ibenthal A, Yohannes D (2020) Survey of deep learning in breast cancer image analysis. Evol Syst 11:5759–5769
- Priyadarshi R, Gupta B (2020) Coverage area enhancement in wireless sensor network. Microsyst Technol 26(5):1417–1426. https://doi.org/10.1007/s00542-019-04674-y
- 111. Kang BJ, Kim MJ, Shin HJ, Moon WK (2022) Acquisition and interpretation guidelines of breast diffusion-weighted MRI (DW-MRI): breast imaging study group of korean society of magnetic resonance in medicine recommendations. Investig Magn Reson. Imaging 26:83–95
- 112. Priyadarshi R, Thakur A, Singh AD (2019) Performance evaluation space-time interest points using branching particle filters. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical Engineering* (Vol. 556, pp. 83–90). Springer Singapore. https://doi.org/10.1007/978-981-13-7091-5_8
- 113. Formaz E et al (2023) Dedicated breast computed-tomography in women with a personal history of breast cancer: a proof-of-concept study. Eur J Radiol 158:110632
- 114. Ren T, Lin S, Huang P, Duong TQ (2022) Convolutional neural network of multiparametric MRI accurately detects axillary lymph node metastasis in breast cancer patients with pre neoadjuvant chemotherapy. Clin Breast Cancer 22:170–177
- Priyadarshi R, Gupta B, Anurag A (2020) Wireless sensor networks deployment: A result oriented analysis. Wireless Pers Commun 113(2):843–866. https://doi.org/10.1007/s11277-020-07255-9
- Boukerroui D, Basset O, Guerin N, Baskurt A (1998) Multiresolution texture based adaptive clustering algorithm for breast lesion segmentation. Eur J Ultrasound 8:135–144
- 117. Priyadarshi R, Kumar RR (2021) An energy-efficient LEACH routing protocol for wireless sensor networks. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical Engineering* (Vol. 673, pp. 423–430). Springer Singapore. https://doi.org/10.1007/978-981-15-5546-6_35
- 118. Corke L et al (2022) Clinical utility of MRI in the neoadjuvant management of early-stage breast cancer. Breast Cancer Res Treat 194:587–595
- 119. Torres-Galván JC et al (2022) Deep convolutional neural networks for classifying breast cancer using infrared thermography. Quant InfraRed Thermogr J 19:283–294

- Wang J, Yang Y (2018) A context-sensitive deep learning approach for microcalcification detection in mammograms. Pattern Recogn 78:12–22
- Deepak S, Ameer PM (2019) Brain tumor classification using deep cnn features via transfer learning. Comput Biol Med 111:103345
- Priyadarshi R, Soni SK, Sharma P (2019) An enhanced GEAR protocol for wireless sensor networks. In V. Nath & J. K. Mandal (Eds.), Lecture Notes in Electrical Engineering (Vol. 511, pp. 289–297). Springer Singapore. https://doi.org/10.1007/978-981-13-0776-8_27
- 123. Liu J et al. (2014) A survey of mri-based brain tumor segmentation methods. Tsinghua Sci Technol 19
- Burgos N, Bottani S, Faouzi J, Thibeau-Sutre E, Colliot O (2021) Deep learning for brain disorders: from data processing to disease treatment. Br Bioinform 22:1560–1576
- 125. (2017) An efficient and automatic glioblastoma brain tumor detection using shift-invariant shearlet transform and neural networks. Int J Imaging Syst Technol 27
- 126. Mittal M et al (2019) Deep learning based enhanced tumor segmentation approach for mr brain images. Appl Soft Comput 78:346–354
- 127. Ali S et al (2020) An effective and improved cnn-elm classifier for handwritten digits recognition and classification. Symmetry (Basel). 12:1742
- Priyadarshi R, Singh MP, Bhardwaj A, Sharma P (2017) Amount of fading analysis for composite fading channel using holtzman approximation. 2017 4th International Conference on Image Information Processing, ICIIP 2017, 2018-Janua, 454–458. https://doi.org/10.1109/ICIIP.2017.83137 59
- 129. Han S, Choi JY (2021) Impact of 18f-fdg PET, PET/CT, and PET/MRI on staging and management as an initial staging modality in breast cancer: a systematic review and meta-analysis. Clin Nucl Med 46:271–282
- 130. Houssami N, Hayes DF (2009) Review of preoperative magnetic resonance imaging (MRI) in breast cancer: should MRI be performed on all women with newly diagnosed, early stage breast cancer? CA Cancer J Clin 59:290–302
- Antunovic L et al (2019) PET/CT radiomics in breast cancer: promising tool for prediction of pathological response to neoadjuvant chemotherapy. Eur J Nucl Med Mol Imaging 46:1468–1477
- Mankoff DA, Sellmyer MA (2022) PET of fibroblast-activation protein for breast cancer diagnosis and staging. (Radiological Society of North America, 2022). https://doi.org/10.1148/radiol. 2021212098.
- 133. Duffy MJ, Walsh S, McDermott EW, Crown J (2015) Biomarkers in breast cancer: where are we and where are we going? Adv Clin Chem 71:1–23
- Hildebrandt MG, Naghavi-Behzad M, Vogsen M (2022) A role of FDG-PET/CT for response evaluation in metastatic breast cancer? Semin Nucl Med 52:520–530
- 135. Bulas D, Egloff A (2013) Benefits and risks of MRI in pregnancy. Semin Perinatol 37:301-304
- 136 Patil RS, Biradar N (2021) Automated mammogram breast cancer detection using the optimized combination of convolutional and recurrent neural network. Evol Intell 14:1459–1474
- 137 Alanazi A (2022) Using machine learning for healthcare challenges and opportunities. Inf. Med Unlocked 30:100924
- Heydarpour F, Abbasi E, Ebadi MJ, Karbassi SM (2020) Solving an optimal control problem of cancer treatment by artificial neural networks. Int J Interact Multimed Art Intell 6(4):18. https:// doi.org/10.9781/ijimai.2020.11.011
- Ghaffari R, Salehi A, Salehi N (2015) Comparison of second molar eruption pattern in skeletal class I and class III malocclusions among 8 9 years old children. Biomed Pharmacol J 8S:811– 816. https://doi.org/10.13005/bpj/788
- Estiri SN, Jalilvand AH, Naderi S, Najafi MH, Fazeli M (2022) A low-cost stochastic computingbased fuzzy filtering for image noise reduction. 2022 IEEE 13th International Green and Sustainable Computing Conference. IGSC 2022:1–6. https://doi.org/10.1109/IGSC55832.2022.9969358
- Esfahani MM, Sadati H (2021) FNIRS signals classification with ensemble learning and adaptive neuro-fuzzy inference system. Proceedings - 2021 7th International Conference on Signal Processing and Intelligent Systems, ICSPIS 2021, 1–5. https://doi.org/10.1109/ICSPIS54653.2021. 9729388
- 142. Wei J, Chammam A, Feng J, Alshammari A, Tehranian K, Innab N, Deebani W, Shutaywi M (2024) Power system monitoring for electrical disturbances in wide network using machine learning. Sustain Comput Inform Syst 42:100959. https://doi.org/10.1016/j.suscom.2024.100959
- Rodrigues AP, Fernandes R, Shetty A, Lakshmanna K, Shafi RM (2022) Real-time twitter spam detection and sentiment analysis using machine learning and deep learning techniques. Comput Intell Neurosci. https://doi.org/10.1155/2022/5211949

- Arevalo J, González FA, Ramos-Pollán R, Oliveira JL, Lopez MAG (2016) Representation learning for mammography mass lesion classification with convolutional neural networks. Comput Methods Prog Biomed 127:248–257
- 145. Roh S, Lee Y-S (2023) Developing culturally tailored mobile web app education to promote breast cancer screening: knowledge, barriers, and needs among American Indian women. J Cancer Educ 2023:1224–123
- 146. Demir, F. (2021) Deepbreastnet: a novel and robust approach for automated breast cancer detection from histopathological images. Biocybern Biomed Eng 41
- 147. Jabeen K et al (2022) Breast cancer classification from ultrasound images using probability-based optimal deep learning feature fusion. Sensors 22:807
- Hirschman J, Whitman S, Ansell D (2007) The black: white disparity in breast cancer mortality: the example of Chicago. Cancer Causes Contr 18:323–33
- Nagore R, Jain PK, Gamad RS, Priyadarshi R (2023) Design of low-power high-efficient singletail comparator using 180 nm CMOS Technology BT - Microelectronics, Communication Systems, Machine Learning and Internet of Things (V. Nath & J. K. Mandal (eds.); pp 155–163). Springer Nature Singapore
- 150. Desai M, Shah M (2021) An anatomization on breast cancer detection and diagnosis employing multi-layer perceptron neural network (mlp) and convolutional neural network (cnn). Clin eHealth 4
- Mokni R, Haoues M (2022) Cadnet157 model: fine-tuned resnet152 model for breast cancer diagnosis from mammography images. Neural Comput Appl 2022:22023–22046
- Swiderski B, Gielata L, Olszewski P, Osowski S, Kołodziej M (2021) Deep neural system for supporting tumor recognition of mammograms using modified gan. Expert Syst Appl 164:113968
- 153. Dhal KG, Ray S, Das A et al (2019) A Survey on nature-inspired optimization algorithms and their application in image enhancement domain. Arch Computat Methods Eng 26:1607–1638. https://doi. org/10.1007/s11831-018-9289-9
- Al-Dhabyani W, Gomaa M, Khaled H, Fahmy A (2020) Dataset of breast ultrasound images. Data Br. 28:104863
- 155. Singh, A. et al. (2021) Ediapredict: an ensemble-based framework for diabetes prediction. ACM Trans Multimed Comput Commun Appl 17
- 156. Ellington TD et al (2023) Trends in breast cancer mortality by race/ethnicity, age, and us census region, United States-1999-2020. Cancer 129:32-38
- SannasiChakravarthy S, Bharanidharan N, Rajaguru H (2022) Multi-deep CNN based experimentations for early diagnosis of breast cancer. IETE J Res 2022:7326–7341
- Madabhushi A, Lee G (2016) Image analysis and machine learning in digital pathology: Challenges and opportunities. Med Image Anal 33:170–175. https://doi.org/10.1016/j.media.2016.06.037
- Sasmal B, Das A, Dhal KG, Ray S (2023) Aquila-particle swarm based cooperative search optimizer with superpixel techniques for epithelial layer segmentation. Appl Soft Comput 149:110947. https:// doi.org/10.1016/j.asoc.2023.110947
- Dhal KG, Das A, Sasmal B, Ray S, Rai R, Garai A (2023) Fuzzy C-Means for image segmentation: challenges and solutions. Multimed Tools Appl 83(9):27935–27971. https://doi.org/10.1007/ s11042-023-16569-2
- Dhal KG, Ray S, Barik S, Das A (2023) Illumination-free clustering using improved slime mould algorithm for acute lymphoblastic leukemia image segmentation. J Bionic Eng 20(6):2916–2934. https://doi.org/10.1007/s42235-023-00392-4
- 162. Dhal KG, Rai R, Das A, Ray S, Ghosal D, Kanjilal R (2023) Chaotic fitness-dependent quasireflected Aquila optimizer for superpixel based white blood cell segmentation. Neural Comput Appl 35(21):15315–15332. https://doi.org/10.1007/s00521-023-08486-0
- 163. Sasmal B, Dhal KG (2023) A survey on the utilization of Superpixel image for clustering based image segmentation. Multimed Tools Appl 82(23):35493–35555. https://doi.org/10.1007/ s11042-023-14861-9
- 164. Ray S, Dhal KG, Kumar Naskar P (2022). Particle swarm optimizer based epithelial layer segmentation in CIElab color space. 7th IEEE International Conference on Recent Advances and Innovations in Engineering, ICRAIE 2022 - Proceedings, 7, 331–336. https://doi.org/10.1109/ICRAIE56454. 2022.10054261
- 165. Dhal KG, Das A, Ray S, Rai R, Ghosh TK (2023) Archimedes optimizer-based fast and robust fuzzy clustering for noisy image segmentation. Journal of Supercomputing 79(4):3691–3730. https://doi.org/10.1007/s11227-022-04769-w
- 166. Dhal KG, Das A, Ray S, Gálvez J (2021) Randomly attracted rough firefly algorithm for histogram based fuzzy image clustering. Knowl-Based Syst 216:106814. https://doi.org/10.1016/j.knosys.2021. 106814

- Iqbal S, Qureshi AN, Aurangzeb K, Alhussein M, Haider SI, Rida I (2023) AMIAC: adaptive medical image analyzes and classification, a robust self-learning framework. Neural Comput Appl. https://doi.org/10.1007/s00521-023-09209-1
- 168. Das A, Dhal KG, Ray S, Gálvez J (2022) Histogram-based fast and robust image clustering using stochastic fractal search and morphological reconstruction. Neural Comput Appl 34(6):4531– 4554. https://doi.org/10.1007/s00521-021-06610-6
- 169. Ray S, Das A, Dhal KG, Gálvez J, Naskar PK (2022) Whale optimizer-based clustering for breast histopathology image segmentation. Int J Swarm Intell Res 13(1):1–29. https://doi.org/10.4018/ IJSIR.302611
- 170. Ray, S., Dhal, K. G., & Naskar, P. K. (2023). Superpixel image clustering using particle swarm optimizer for nucleus segmentation. In M. Thakur, S. Agnihotri, B. S. Rajpurohit, M. Pant, K. Deep, & A. K. Nagar (Eds.), *Lecture Notes in Networks and Systems* (Vol. 547, pp 445–457). Springer Nature Singapore. https://doi.org/10.1007/978-981-19-6525-8_34
- 171. Dhal KG, Rai R, Das A, Ghosh TK (2022). Hybridization of Sine-cosine algorithm with k-means for pathology image clustering. In: A. A. Sk, T. Turki, T. K. Ghosh, S. Joardar, & S. Barman (Eds.), *Communications in Computer and Information Science: Vol. 1695 CCIS* (pp. 76–86). Springer Nature Switzerland. https://doi.org/10.1007/978-3-031-22485-0_8
- 172. Podda AS et al (2022) Fully-automated deep learning pipeline for segmentation and classification of breast ultrasound images. J Comput Sci 63:101816
- 173. Nicolas E, Khalifa N, Laporte C, Bouhroum S, Kirova Y (2021) Safety margins for the delineation of the left anterior descending artery in patients treated for breast cancer. Int J Radiat Oncol Biol Phys 109:267–272
- 174. Shen T, Wang J, Gou C, Wang FY (2020) Hierarchical fused model with deep learning and type-2 fuzzy learning for breast cancer diagnosis. IEEE Trans Fuzzy Syst 28:3204–3218
- 175. Hodkinson A et al (2022) Associations of physician burnout with career engagement and quality of patient care: systematic review and meta-analysis. BMJ 378:e070442
- 176. Simsek A et al. (2021) Factors affecting the accuracy of 18 f-FDG PET/CT in detecting additional tumor foci in breast cancer. Arch Hell Med/Arheia Ellenikes Iatrikes 38
- 177. Kooi T et al (2016) Large scale deep learning for computer aided detection of mammographic lesions. Med Image Anal 35:303–312
- 178. Charbonnier J et al (2017) Improving airway segmentation in computed tomography using leak detection with convolutional networks. Med Image Anal. 36:52–60
- 179. Bejnordi BE, Veta M, Diest PJ (2017) Diagnostic assessment of deep learning algorithms for detection of lymph node metastases in women with breast cancer. JAMA 318:2199–2210
- 180. Karimi D, Samei G, Kesch C, Nir G, Salcudean SE (2018) Prostate segmentation in mri using a convolutional neural network architecture and training strategy based on statistical shape models. Int J Comput Assist Radiol Surg 13:1211–1219
- 181. Esteva A. et al. (2017) Dermatologist-level classification of skin cancer with deep neural networks. Proc Nat 542
- Anwar SM et al (2018) Medical image analysis using convolutional neural networks: a review. J Med Syst 42:226
- 183. Kennard K et al (2022) Outcomes of abbreviated MRI (ab-MRI) for women of any breast cancer risk and breast density in a community academic setting. Ann Surg Oncol 29:6215–622
- Havaei M, Davy A, Warde-Farley D (2017) Brain tumor segmentation with deep neural networks. Med Image Anal. 35:18–31
- 185. Kamnitsas K et al (2017) Efficient multi-scale 3d CNN with fully connected CRF for accurate brain lesion segmentation. Proc Med Image Anal 36:61–78
- 186. Priyadarshi R (2024) Exploring machine learning solutions for overcoming challenges in IoTbased wireless sensor network routing: a comprehensive review. Wireless Netw. https://doi.org/10. 1007/s11276-024-03697-2
- 187. Kooi T et al (2017) Large scale deep learning for computer aided detection of mammographic lesions. Med Image Anal 35:303–312
- Cheng H, Jiang X, Sun Y, Wang J (2001) Color image segmentation: advances and prospects. Pattern Recogn. 34:2259–2281
- Pan Z, Lu J (2007) A bayes-based region-growing algorithm for medical image segmentation. Comput Sci Eng. 9:32–38
- 190. Kallenberg M et al (2016) Unsupervised deep learning applied to breast density segmentation and mammographic risk scoring. Proc IEEE Trans Med Imaging 35:1322–1331

- 191. Pereira S et al (2018) Enhancing interpretability of automatically extracted machine learning features: application to a rbm-random forest system on brain lesion segmentation. Med Image Anal. 44:228–244
- 192. Singh MP, Priyadarshi R, Garg P (2019) Design of SIW-fed broadband microstrip patch antenna for E-band wireless communication. In A. K. Luhach, K. B. G. Hawari, I. C. Mihai, P.-A. Hsiung, & R. B. Mishra (Eds.), *Smart Computational Strategies: Theoretical and Practical Aspects* (pp. 185–193). Springer Singapore. https://doi.org/10.1007/978-981-13-6295-8_16
- 193. Porwal P et al (2018) Indian diabetic retinopathy image dataset (idrid): a database for diabetic retinopathy screening research. MDPI Data 3:25
- Setio AAA, Jacobs C, Gelderblom J, Ginneken B (2015) Automatic detection of large pulmonary solid nodules in thoracic CT images. Med Phys. 42:5642–5653
- 195. Ginneken B, Stegmann M, Loog M (2006) Segmentation of anatomical structures in chest radiographs using supervised methods: a comparative study on a public database. Med Image Anal. 10:19–40
- Ye M, Giannarou S, Meining A, Yang G-Z (2015) Online tracking and retargeting with applications to optical biopsy in gastrointestinal endoscopic examinations. Med Image Anal. 30:144–157
- Qiu Y, Ma L, Priyadarshi R (2024) Deep learning challenges and prospects in wireless sensor network deployment. Arch Comput Methods Eng. https://doi.org/10.1007/s11831-024-10079-6
- 198. Li T et al (2023) A systematic review of the impact of the covid-19 pandemic on breast cancer screening and diagnosis. Breast 67:78–88
- Yu L, Chen H, Dou Q, Qin J, Heng P-A (2016) Automated melanoma recognition in dermoscopy images via very deep residual networks. IEEE Trans Med Imaging. 36:994–1004
- Oktay O et al (2018) Anatomically constrained neural networks (ACNN): application to cardiac image enhancement and segmentation. IEEE Trans Med Imaging 37:384–395
- Alex V, Vaidhya K, Thirunavukkarasu S, Kesavadas C, Krishnamurthia G (2017) Semisupervised learning using denoising autoencoders for brain lesion detection and segmentation. J Med Imaging. 4:041311
- Singh MP, Priyadarshi R, Sharma P, Thakur A (2017). Small size rectangular microstrip patch antenna with a cross slot using SIW. 2017 4th International Conference on Image Information Processing, ICIIP 2017, 2018-Janua, 446–449. https://doi.org/10.1109/ICIIP.2017.8313757
- 203. Avendi MR, Kheradvar A, Jafarkhani H (2016) Automatic segmentation of the right ventricle from cardiac MRI using a learning-based approach. Magn Reson Med. 78:2439–2448
- Azizi S et al (2016) Detection of prostate cancer using temporal sequences of ultrasound data: a large clinical feasibility study. Surgery 11:947–95
- Shorten C, Khoshgoftaar TM (2019) A survey on image data augmentation for deep learning. J Big Data 6:60
- 206. Srivastava N, Hinton G, Krizhevsky A, Sutskever I, Salakhutdinov R (2014) Dropout: a simple way to prevent neural networks from overfitting. J Mach Learn Res 15:1929–1958
- 207. Priyadarshi R, Singh MP, Tripathi H, Sharma P (2017) Design and performance analysis of vivaldi antenna at very high frequency. 2017 4th International Conference on Image Information Processing, ICIIP 2017, 2018-Janua, 450–453. https://doi.org/10.1109/ICIIP.2017.8313758
- Thakur RS, Yadav RN, Gupta L (2019) State-of-art analysis of image denoising methods using convolutional neural networks. IET Image Proc 13:2367–2380
- Talebi H, Zhu X, Milanfar P (2013) How to saif-ly boost denoising performance. IEEE Trans Image Process 22:1470–1485
- Yi X, Walia E, Babyn P (2019) Generative adversarial network in medical imaging: a review. Med Image Anal 58:101552
- Priyadarshi R, Vikram R (2023) A triangle-based localization scheme in wireless multimedia sensor network. Wireless Pers Commun 133(1):525–546. https://doi.org/10.1007/ s11277-023-10777-7
- 212. Song Y, Zhu Y, Du X (2019) Dynamic residual dense network for image denoising. Sensors 19:3809
- Priyadarshi R (2024) Energy-efficient routing in wireless sensor networks: A meta-heuristic and artificial intelligence-based approach: A comprehensive review. Arch Computat Methods Eng. https://doi.org/10.1007/s11831-023-10039-6
- Wang F, Henninen TR, Keller D, Erni R (2020) Noise2atom: unsupervised denoising for scanning transmission electron microscopy images. Appl Microsc 50:23
- 215. Ahmad AJ, Hassan SD, Priyadarshi R, Nath V (2023) Analysis on image compression for multimedia communication using hybrid of DWT and DCT. In V. Nath & J. K. Mandal (Eds.), *Lecture Notes in Electrical Engineering* (Vol. 887, pp. 667–672). Springer Nature Singapore. https://doi. org/10.1007/978-981-19-1906-0_54

- 216. Xiang L et al (2017) Deep auto-context convolutional neural networks for standard-dose pet image estimation from low-dose PET/MRI. Neurocomputing 267:406–416
- Weigert M et al (2018) Content-aware image restoration: pushing the limits of fluorescence microscopy. Nat Methods 15:1091–1097
- Ulaner GA (2019) PET/CT for patients with breast cancer: where is the clinical impact? Am J Roentgenol 213:254–265
- 219. Hassan NM, Hamad S, Mahar K (2022) Mammogram breast cancer cad systems for mass detection and classification: a review. Multimed Tools Appl 81:20043–20075
- Sateesh VA, Dutta I, Priyadarshi R, Nath V (2021) Fractional frequency reuse scheme for noise-limited cellular networks BT - Proceedings of the Fourth International Conference on Microelectronics, Computing and Communication Systems (V. Nath & J. K. Mandal (eds.); pp. 995–1004). Springer Singapore
- Randheer Soni SK, Kumar S, Priyadarshi R (2020) Energy-aware clustering in wireless sensor networks BT - Nanoelectronics, Circuits and communication systems (V. Nath & J. K. Mandal (eds.); pp. 453–461). Springer Singapore
- 222. Jiang J et al (2022) Breast cancer detection and classification in mammogram using a three-stage deep learning framework based on paa algorithm. Artif Intell Med 134:102419
- 223. Saha A et al (2018) A machine learning approach to radiogenomics of breast cancer: a study of 922 subjects and 529 dce-MRI features. Br J Cancer 119:508–516
- 224. Nagalakshmi T (2022) Breast cancer semantic segmentation for accurate breast cancer detection with an ensemble deep neural network. Neural Process Lett 54:5185–5198
- 225. Priyadarshi R, Soni SK, Bhadu R, Nath V (2018) Performance analysis of diamond search algorithm over full search algorithm. Microsyst Technol 24(6):2529–2537. https://doi.org/10.1007/s00542-017-3625-0
- 226. Rawat P, Chauhan S, Priyadarshi R (2021) A novel heterogeneous clustering protocol for lifetime maximization of wireless sensor network. Wireless Pers Commun 117(2):825–841. https://doi.org/10. 1007/s11277-020-07898-8
- Kumar RR, Kumar A, Srivastava S (2020) Anisotropic diffusion based unsharp masking and crispening for denoising and enhancement of MRI images. 2020 International Conference on Emerging Frontiers in Electrical and Electronic Technologies (ICEFEET), Patna, India, pp 1–6. https://doi.org/ 10.1109/ICEFEET49149.2020.9186966
- Evrimler S, Algin O (2021) CT and MR enterography and enteroclysis BT Medical imaging contrast agents: a clinical manual. in (eds. Erturk, S. M., Ros, P. R., Ichikawa, T. & Saylisoy, S.) (Springer, 2021). https://doi.org/10.1007/978-3-030-79256-5_14.
- 229. Singh S, Kumar R (2022) Breast cancer detection from histopathology images with deep inception and residual blocks. Multimed Tools Appl 81:5849–5865

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Springer Nature or its licensor (e.g. a society or other partner) holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.